

NAVAL POSTGRADUATE SCHOOL Monterey, California







THESIS

MISSIONING WITH MINIMAL DELAYED ENTRY PROGRAM (DEP) LOSS

by

Bryan D. Burris

September, 1993

Thesis Advisor:

Siriphong Lawphongpanich

Approved for public release; distribution is unlimited.



94 1 28 003

Unclassified

Security Classification of this page

	REPORT DOCI	UMENTATION PAGE					
la Report Security Classification: Unclassif	ied	1b Restrictive Markings					
2a Security Classification Authority		3 Distribution/Availability of	of Report				
2b Declassification/Downgrading Schedule		Approved for public re	lease; distri	bution is u	nlimited.		
4 Performing Organization Report Number(s)		5 Monitoring Organization	Report Numbe	:f(s)			
6a Name of Performing Organization Naval Postgraduate School	6b Office Symbol (if applicable) *OR	7a Name of Monitoring Org Naval Postgraduate Sch	zanization hool				
6c Address (city, state, and ZIP code) Monterey CA 93943-5000	7b Address (city, state, and Monterey CA 93943-5	21P code) 000					
8a Name of Funding/Sponsoring Organization	9 Procurement Instrument Identification Number						
Address (city. state, and ZIP code)		10 Source of Funding Numbers					
		Program Element No Project No Task No Work Unit Accession No					
11 Title (include security classification) MIS	SIONING WITH MININ	AAL DELAYED ENTRY	PROGRA	M (DEP) L	.OSS		
12 Personal Author(s) Bryan D. Burris							
13a Type of Report Master's Thesis	13b Time Covered From To	14 Date of Report (year. month. day)15 Page Count9823 Sep 93					
16 Supplementary Notation The views exp the Department of Defense or the U.S	ressed in this thesis are . . Government.	those of the author and d	o not reflec	t the offici	al policy or position of		
17 Cosati Codes	18 Subject Terms (con	Vinue on reverse if necessary of	and identify by	block numb	er)		
Field Group Subgroup Contract Missioning, Delayed Entry Program, DEP Loss, United States Army Recruiting Command, USAREC							
19 Abstract (continue on reverse if necessary This thesis addresses the mist	and identify by block number, sioning problem which is) s to determine the number	r of individu	uals to be r	recruited or contracted		

each month by Army recruiters. After signing the contracts, these individuals are enrolled in the Delayed Entry Program (DEP) prior to their basic training. During DEP, some individuals may renege on their contracts, thus becoming DEP losses. Although DEP is costly, it is necessary since it acts as an inventory of recruits to smooth out the training loads at boot camps and allows the Army to perform a background check for each recruit.

The missioning problem is formulated as a linear program that minimizes the expected DEP loss subject to requirements imposed by the Deputy Chief of Staff for Personnel (DCSPER) and U.S. Army Recruiting Command (USAREC). Integral to the formulation are the estimates of DEP loss probabilities for various combinations of recruit categories and DEP durations. The estimates are based on a Binomial assumption and Isotonic regression.

The linear programming model of the missioning problem is implemented in GAMS and provides results indicating that DEP loss can be reduced from the current level of 11.46% to 8.59%. This translates to nearly \$11 million saving annually. Other applications of the model are also provided.

20 Distribution/Availability of Abstract <u>xx</u> unclassified/unlimitedsame as report	_ DTIC users	21 Abstract Security Classification Unclassified	
22a Name of Responsible Individual Siriphong Lawphongpanich		22b Telephone (include Area Code) (408) 656-2106	22c Office Symbol OR/Lp
DD FORM 1473,84 MAR	83 APR edition m	nay be used until exhausted	security classification of this page
	All other e	ditions are obsolete	Unclassified

Approved for public release; distribution is unlimited.

Missioning with Minimal Delayed Entry Program (DEP) Loss

by

Bryan D. Burris Captain, United States Army B.S., University of Georgia

Submitted in partial fulfillment of the requirements for the degree of

MASTER OF SCIENCE IN OPERATIONS RESEARCH

from the

NAVAL POSTGRADUATE SCHOOL

September, 1993

Author:

Bya D. Bi

Bryan D. Burris

Approved by:

anulonasanie 3/1mm Siriphong Lawphongpanich, Thesis Advisor

Whitaker, Second Reader

Peter Purdue, Chairman Department of Operations Research

ABSTRACT

This thesis addresses the missioning problem which is to determine the number of individuals to be recruited or contracted each month by Army recruiters. After signing the contracts, these individuals are enrolled in the Delayed Entry Program (DEP) prior to their basic training. During DEP, some individuals may renege on their contracts, thus becoming DEP losses. Although DEP is costly, it is necessary since it acts as an inventory of recruits to smooth out the training loads at boot camps and allows the Army to perform a background check for each recruit.

The missioning problem is formulated as a linear program that minimizes the expected DEP loss subject to requirements imposed by the Deputy Chief of Staff for Personnel (DCSPER) and U.S. Army Recruiting Command (USAREC). Integral to the formulation are the estimates of DEP loss probabilities for various combinations of recruit categories and DEP durations. The estimates are based on a Binomial assumption and lsotonic regression.

The linear programming model of the missioning problem is implemented in GAMS and provides results indicating that DEP loss can be reduced from the current level of 11.46% to 8.59%. This translates to nearly \$11 million saving annually. Other applications of the model are also provided.

A00695	ton For	
BTIS	OFALI	R
DTIC T	4B	
Vuann₀	anova 👘	n
Justia	listicn.	
Avr.13	ability	J <u>2</u> 483
······	tvol) en	
Dist	\$.9018	£
	1	
0~1		
r		

iii

DTIC QUALITY INSPECTED 2

TABLE OF CONTENTS

.

		•	
I.	INT	RODUCTION	1
	A.	BACKGROUND	1
	Β.	THESIS OBJECTIVE	3
	c.	THESIS ORGANIZATION	3
II.	RE	CRUITING PROCESS AT USAREC	4
	A.	USAREC	4
		1. USAREC Organization	4
		2. Interactions With DCSPER	5
	B.	MISSIONING PROCESS AT USAREC	7
	c.	RECRUITING PROCESS	10
III	. 0	PTIMAL MISSIONING MODEL	12
	A.	PLANNING RECRUITING MISSIONS	13
		1. DA Requirements	13
		a. Volume Requirement	13
		b. PS/NPS Mix Requirement	13
		c. Quality Requirement	14
		d. Jender Requirement	14
		2. Operational Requirements	14
		a. Recruiter Workload	14
		b. Monthly Accession Limitation	14

		c. Proportion of Annual Contracts Written	
		Each Month	15
		d. Time in DEP Requirement	16
		e. HSDG NPS-A Quality Mix Requirement	16
		3. Future Accession Requirements	17
		a. Volume Requirement	17
		b. Quality Requirement	17
	в.	PROBLEM STATEMENT	18
	c.	PROBLEM FORMULATION	18
IV.	ES'	TIMATING DEP LOSS PROBABILITIES	27
	A.	DESCRIPTION OF DATA	28
	В.	PRELIMINARY ANALYSIS	29
	c.	DEP LOSS PROBABILITY BY QUARTER	33
	D.	ESTIMATING DEP LOSS PROBABILITIES	35
v.	MOD	EL IMPLEMENTATION AND ANALYSIS	39
	A.	DATA INPUTS	39
	в.	MODEL OUTPUTS	41
	c.	SAMPLE ANALYSES	43
		1. Female Proportion	44
		2. Contracts Produced in FY94 for FY95	
		Accession	45
VI.	CO	NCLUSION	47
	A.	SUMMARY	47

В.	ARE	AS FO	DR FU	TURE	E RE	ESEA	ARC	н.	•	•	•	•	٠	•	٠	•	•	•	48
	1. (Condi	ition	al I	DEP	Los	s	Pro	bal	oi]	lit	ie	s	•	•	•	•	•	48
	2. N	lissi	ionin	g by	, Bi	riga	ade	•	•	•	•	•	•	•	•	•	•	•	49
APPENDI	ХА	CHI	-SQUA	RED	TES	ST	•	•••	•	•	•	•	•	•	•	•	•	•	50
APPENDI	хв	DEP	LOSS	PRC	BAB	BILI	[TY	ES	TI	MAT	res	;		•	•	•	•	•	64
APPENDI	хс9	5% C(ONFIE	DENCE	E IN	ITEF	(VA	LU	PPE	R '	IAV	LUE	E P	RC	BA	BI	LI	TY	
EST	IMATI	ES .	• •	• •	•		•	• •	•	•	•	•	•	•	•	•	•	•	70
	Y D	FOR		א השי	15m	ספר	MA	י בית	v										76
AFFENDI.		PORI	SCAD I		41211		1.1.12	INI	Λ	•	٠	•	•	•	•	•	•	•	70
APPENDI	ХE	•••	•••	• •	•		•	•••		•	•	•	•	•	•	•	•		77
LIST OF	REFI	EREN	CES	• •	•	•••	•	•••	•	•	•	•	•	•	•	•	•	•	83
ተለተጥተል፣	חזפי	זפזקי	ነጥተ ∩ •	1 1.74	ናጥ														Q /
THT 1 TUD	010					• •	•	• •	•	•	•	•	•	•	•	•	•	•	04

LIST OF TABLES

Table I 1994 DA ACCESSION MISSION	6
Table II MISSION BOXES	8
Table III MISSION BOXES VS. RECRUITING CATEGORIES	8
Table IV DEFINITION OF QUARTERS	9
Table V MONTHLY ACCESSION REQUIREMENT	15
Table VI MINIMASTER FILES FROM FY88-FY92	28
Table VII NUMBER OF RECORDS BY YEAR AND QUARTER	29
Table VIII EXAMPLE DATA NEEDED FOR CHI-SQUARED TEST .	31
Table IX OBSERVATIONS USED FOR ANALYSIS	32
Table X QUARTERLY ACCESSIONS AND DEP LOSS	33
Table XI RESULTS FROM ISOTONIC REGRESSION FOR GF4	38
Table XII USER DEFINED PARAMETER VALUES	40
Table XIII RECOMMENDED CONTRACT MISSION FOR QTR1, FY94	42
Table XIV NPS FEMALE SENSITIVITY ANALYSIS	45
Table XV FY95 NPS-A ACCESSIONS, CONTRACTED DURING FY94	46

vii

LIST OF FIGURES

.

Figure 1	Organizational Chart	•	•	•	•	4
Figure 2	Preliminary DEP Loss Probabilities .	•	•	•	•	29
Figure 3	Quarterly DEP Loss Probability	•	•	•	•	34
Figure 4	First Quarter DEP Loss For GMA	•	•	•	•	36
Figure 5	Male vs. Female DEP Loss Probability	•		•	•	44

.

ACKNOWLEDGEMENTS

This thesis is the result of the combined effort and support of a number of very important people.

First, I owe a great deal of love and appreciation to my wife for understanding the requirements of this thesis. She served as a supporting wife, and both father and mother for our children during the last 24 months.

Second, I owe hours of make-up "fun time" to my children. They were forever patient, knowing that I had a job to do, and that their cooperation would make it easier.

Third, special thanks go to my thesis advisor, DR. Siriphong Lawphongpanich, whose absolutely untiring effort and support went well beyond what was expected. His technical and literary expertise made the thesis experience enjoyable and educational. Fourth, I express my gratitude to DR. Lyn Whitaker. Her advise was critical in completing the statistical requirements of this thesis.

Finally, I would like to thank MAJ John Hershberger and CPT Dan Bunning of the United States Army Recruiting Command (USAREC). MAJ Hershberger's support and subject matter expertise were invaluable. He was always available for moral support and answers needed to improve the product. CPT Bunning saved months of thesis effort by collecting and screening the USAREC data needed to complete this thesis.

ix

EXECUTIVE SUMMARY

A. BACKGROUND

The United States Army Recruiting Command (USAREC) is responsible for recruiting civilians to enlist into the Army. The Department of the Army specifies the number (or volume) and quality of enlistments that USAREC must enlist annually.

To accomplish the task as specified by DA, Army recruiters must perform duties like a salesman, i.e., they must sell the Army to American youths between the ages of 17 and 21. These recruiters begin by contacting youths at high schools, at recruiting stations or through some informal introduction. If a youth is interested and eligible to join the Army, he or she is processed for enlistment at a Military Entrance Processing Station (MEPS). At this point, the youth usually selects a career and signs a contract agreeing to join the Army.

After signing the contract to join the Army, a recruit must wait from 1 to 12 months before he or she enters basic training. During this period, the recruit becomes a member of the Delayed Entry Program (DEP). The DEP serves two main purposes, it allows the Army to complete required background checks on the recruit and maintain a level training load at basic training locations.

х

While in DEP, recruits are left with time to pursue other career opportunities. In general, the longer a recruit remains in DEP, the higher the possibility that he or she will renege on the enlistment contract. When this occurs, the recruit becomes a DEP loss. USAREC estimates that approximately 15% of the enlistment contracts become DEP losses. Based on an estimated cost of \$5000 per recruit and 13,400 annual DEP losses, the Army estimates it spends approximately \$67 million each year on DEP loss, a during the current budget considerable sum cutting environment.

B. PROBLEM STATEMENT

On a quarterly basis, USAREC analysts must set monthly recruiting goals for each of its four recruiting brigades. The quotas simply specify the number of contracts each brigade must enlist each month. To set these quotas, the analysts must consider three main factors: the DA recruiting requirements, the USAREC recruiting policies and the expected DEP loss. To account for the expected DEP loss, USAREC simply increases the DA mission by 15%. Although this method is offective, it does not account for the differences in expected DEP loss for the different groups (i.e., female, male, high school graduate, non high school grad, etc.) within the recruit population. Thus, the goal of this thesis is to

xi

develop a method which accounts for DEP loss more efficiently in setting the monthly recruiting quotas.

C. APPROACH

The problem of setting the monthly recruiting quotas involves two areas of operations research: statistics and optimization. Statistics is used to estimate the probability a recruit will become a DEP loss. Given these probabilities, the problem of setting the recruiting quota is formulated as an optimization problem with the objective of minimizing DEP losses.

D. ACCOMPLISHMENTS

In achieving the above goal, this thesis made two major accomplishmencs. One is the estimation of the probability of a recruit becoming a DEP loss. The Army groups recruits into 22 classifications based on their gender, Armed Forces Qualification Test (AFQT) score, education and whether or not they have had pricr military service. The probability of DEP loss depends on these 22 classifications along with the length of time a recruit remains in the DEP. Based on the recruiting data from 1988 to 1992, this thesis provides estimations of DEP loss for each combination of classifications and DEP length. By themselves, these probabilities provide USAREC with information useful for setting recruiting strategies at all levels of command, i.e.. from the headquarters down to the

xii

recruiters. Furthermore, they also identify classifications that should be avoided if expenses are to be kept to a minimum.

The other accomplishment deals with an application of these probabilities in setting the monthly recruiting mission at the headquarters level. This thesis formulates the problem of setting monthly recruiting missions as an optimization problem with the objective of minimizing the number of DEP losses. The main data for this problem are the DEP loss probabilities, for they determine the expected number of DEP losses. The optimization model is implemented in the General Algebraic Modeling System (GAMS). Preliminary results indicate that the model facilitates the process of setting the monthly recruiting mission as well as allows for analysis of various recruiting policies.

I. INTRODUCTION

A. BACKGROUND

In support of ongoing cuts in the Army's budget, the United States Army Recruiting Command (USAREC) seeks various ways to reduce its operating cost. Despite this cost cutting effort, USAREC must continue to recruit quality young men and women to meet the future needs of the Army.

Among the many areas with high operating cost, USAREC is concerned with the rising cost due to the loss of recruits prior to being sent off to basic training. After signing a contract to enter the Army, a recruit must wait from 1 to 12 months before he or she enters basic training [Ref. 1]. During this period, the recruit becomes a member of the *Delayed Entry Program (DEP)*. While in the DEP, some recruits decide not to join the Army and thus renege on their contract. This situation is referred to as *DEP loss*.

USAREC estimates that 15% of the DEP population will become a DEP loss prior to entering basic training. The Army estimates that it costs approximately \$5000 to recruit an individual and there are approximately 13,400 DEP losses annually. This translates to \$67 million the Army must spend on DEP loss alone [Ref. 2]. This sum is considerable during the current budget environment.

There are several alternatives to reduce the cost of DEP loss. First, USAREC could increase the required contact time between recruiters and their DEP recruits. These contacts include phone conversations and personal contacts in the form of social or Army related activities. These events are designed to maintain a recruit's interest in the Army and provide the recruiter with the setting to discuss the recruit's contractual obligation along with the continued career opportunities that exist in the Army. Second, the Army could exercise its legal right, forcing recruits to honor their enlistment contract. The contract is a legal document binding both parties to the terms therein. To date, the Army has not chosen to exercise this option. And finally, USAREC could incorporate expected DEP losses into their operational In this case, a recruit is viewed as a commodity planning. which is perishable with time, and recruits in DEP serve as an inventory to fulfill the Army's future demand for soldiers. However, the longer the recruits remain in DEP, the more likely they will become a DEP loss. This suggests a just-intime inventory policy, which is not implementable for two reasons. First, recruiting is not a deterministic process. The number of recruits can vary greatly from month to month. Second, a minimum of one month DEP is required for performing a required background check on each recruit. Thus, a DEP is a necessary part of recruiting and should be efficiently managed.

B. THESIS OBJECTIVE

This thesis focuses on the last alternative to reduce the cost due to DEP loss. In particular, this thesis considers the problem of managing DEP loss as an optimization model and uses statistical techniques to estimate inputs to the model. One important input requiring statistical estimation is the probability of a recruit becoming a DEP loss. This probability depends on the characteristics of the recruit and the length of time he or she must spend in the DEP.

C. THESIS ORGANIZATION

Chapter II introduces the recruiting process within USAREC. Chapter III describes the optimization model. The model provides the best combination of recruit characteristics and DEP length that satisfy the needs of the Army. Chapter IV describes the statistical techniques used to estimate the DEP loss probability associated with each recruit characteristic and DEP length combination. It also studies the variation of DEP loss probabilities on a quarterly basis. Chapter V discusses the implementation of the optimization model. Finally, Chapter VI summarizes the results of the thesis and presents recommendations for future study.

II. RECRUITING PROCESS AT USAREC

The sections below provide information concerning the recruiting process at USAREC. The first section describes the organizational structure at USAREC and explains its interaction with the Deputy Chief of Staff for Personnel (DCSPER). The second section discusses the planning (or missioning) process at USAREC Headquarters (HQ). The last section describes the 20 miting process at USAREC's lower organizational level:

A. USAREC

1. USAREC Organization

At the top most level of the Army recruiting organization is the USAREC Headquarters (HQ), which consists of ten directorates (see Figure 1).



Figure 1 Organizational Chart

Under the direction of the Commanding General, these directorates coordinate and support the recruiting effort for the entire U.S. Army. The next level in the organization consists of four brigades, each responsible for recruiting in one of four separate geographical regions of the continental United States (CONUS). Under each brigade is a collection of battalions which are generally responsible for recruiting in one or two states. Then, there is a collection of companies in each battalion and, similarly, a collection of stations in each company. The recruiters producing enlistment contracts (i.e., On-Line Production Regular Army (OPRA) recruiters) are located at the stations. The OPRA recruiters form the sales force for the Regular Army (RA). The Army Reserve's mission is recruited by USAREC Reserve Recruiters working alongside the RA recruiters. This thesis only addresses the RA recruiting effort.

Currently, USAREC contains a total of 4 brigades, 42 battalions, 222 companies, 1495 stations and 4200 OPRA recruiters. These numbers are expected to change in the near future due to the ongoing Army force reduction and realignment process.

2. Interactions With DCSPER

It is the Directorate of Personnel Management at DCSPER, who establishes the Army's future personnel needs and sets the Department of the Army (DA) accession mission for

USAREC. The DA accession mission specifies the number of individuals who must access into or enter the Army as well as the proportion of recruits in various categories. These categories group recruits by their service history, education level, test score on the Armed Forces Qualification Test (AFQT) and gender. Table I summarizes the accession mission for 1994.

Table I 1994 DA ACCESSION MISSION

Total Accessions (Volume) = 75,000 Service Mix = 70,000 Non-Prior Service (NPS) = 5,000 Prior Service (PS) Quality Mix for NPS accessions ≥ 95% must be high school graduates (HSDG) ≥ 67% must score in the top 50th percentile on the AFQT (NPS-A) ≤ 2% can score between the 21st and 30th percentile on the AFQT (TSC-4)* Gender Mix for NPS accessions ≥ 14.8% must be female

* Current policy restricts TSC-4 to scores between the 26th and 30th percentile.

From Table I, there are six categories of interest. They include recruits (i) with prior military service (PS), (ii) with no prior military services (NPS), (iii) with a high school diploma (HSDG), (iv) with no prior service and an AFQT score in the top 50th percentile (NPS-A), (v) with no prior service and an AFQT score between the 26th and 30th percentiles (TSC-4) and (vi) female (FEM) recruits.

B. MISSIONING PROCESS AT USAREC

The analysts at the Mission Division of the Program Analysis and Evaluation (PAE) Directorate are responsible for insuring that USAREC fulfills the DA accession mission. To do so, the analysts must set a quarterly recruiting goal that specifies the number of individuals (or contracts) to be recruited during each month of the quarter. This goal is referred to as the *contract mission*. Since the analysts must account for DEP losses, the number of contracts produced annually is typically larger than the number of accessions required by DCSPER.

To insure that the accessions contain the required proportions of various DCSPER recruit categories, the analysts first group the recruits (or contracts) into 22 mission BOXES or classifications according to their service history, education level, AFQT score and gender. Below, Table II provides a description of the 22 mission BOXES and Table III specifies how each BOX contributes to each of the six DCSPER recruit categories described in Section A. Note that category HSDG (High School Degree Graduate) in Table III includes high school seniors, since they are expected to graduate before they depart for basic training.

MISSI	ION BO	<u>Kes</u>						
SMA	GMA	SMB	GMB	SM4	GM4	NMA	NMB	
SFA	GFA	SFB	GFB	SF4	GF4	NFA	NFB	
MPS	FPS	HMA	HMB	HFA	HFB			
<u>I,EGEN</u>	<u>VD</u>							
HIGH	SCHOOL	L EDUC	ATION:					
G	= Gra	advate	N	= Nor	n-Grad	uate		
S	= Ser	nior	Н	= Ger	neral 🗄	Equival	lency	Diploma
SERVI	ICE HIS	STORY :				-	-	-
PS	= Pr:	ior Sea	rvice	NPS	5 = Not	n-Prio	r Serv	vice
GENDE	ER:							
М	= Mal	le	F	= Fer	nale			
AFQT	SCORE	:						
Α	= 50t	th Pere	centile	e or be	etter			
В	= bet	tween 🔅	31st ar	nd 49tl	n Perc	entile		
4	= bet	tween (26th ar	nd 30tl	n Perc	entile		

1.4

Table III MISSION BOXES VS. RECRUITING CATEGORIES

BOX SMA G. 'A SMB	<u>PS</u>	<u>NPS</u> YES YES YES	<u>HSDG</u> YES YES YES	<u>NPS-A</u> YES YES	<u>TSC-4</u>	<u>Fem</u>	
GMB		YES	YES		VEC		
GM4		IES	IES	VDO	TES		
NMA		IES		IES			
NMB		YES					
SFA		YES	YES	YES		YES	
GFA		YES	YES	YES		YES	
SFB		YES	YES			YES	
GFB		YES	YES			YES	
GF4		YES	YES		YES	YES	
NFA		YES		YES		YES	
NFB		YES				YES	
MPS	YES						
FPS	YES						
HMA		YES		YES			
HMB		YES					
HFA		YES		YES		YES	
HFB		YES				YES	
Note:	USAR	EC cu	rrently	not re	ecruiting	SM4 and	SF4

.

The next step requires dividing the monthly recruiting quotas into the number of contracts to be recruited within each mission BOX. In general, the task of setting the recruiting quotas by mission BOX is time consuming and requires analysts with considerable experience in recruiting. Ultimately, the monthly recruiting mission must produce the desired number of acc⁷ ssions with the desired mix of recruit categories. This requires the analysts to closely monitor the progress of recruiting on a continual basis in order to effect any necessary adjustments to the recruiting quotas.

On a quarterly basis, the monthly recruiting quotas are divided and assigned, by mission BOX, to each of the four brigades. For future reference, Table IV defines each of the quarters.

OUARTER	SEASON	MONTHS
1	Fall	Oct/Nov/Dec
2	Winter	Jan/Feb/Mar
3	Spring	Apr/May/Jun
4	Summer	Jul/Aug/Sep

Table IV DEFINITION OF QUARTERS

Given the time required to distribute the mission throughout the recruiting organization, analysts typically set the quarterly quotas at least one quarter in advance. This gives the brigades three to five months to distribute the mission to their battalions. Each battalion will do the same for its companies and so on until the mission finally reaches the recruiters. On average, a recruiter is expected to produce between one and two contracts per month.

C. RECRUITING PROCESS

To produce a contract in a given month, recruiters perform duties much like a salesman. In fact, they sell the Army benefits to American youths predominately between the ages of 17 and 21. These recruiters begin by making contacts with potential recruits at high schools, recruiting stations or through informal introductions. Once these young men and women become interested in joining the Army, they are given a physical and mental (AFQT) test to determine eligibility. Any AFQT score below 26 is an automatic ineligibility.

If the prospect is eligible to join the Army, he or she is processed for enlistment at a Military Entrance Processing Station (MEPS). It is at the MEPS where the prospect usually selects a career path and signs a contract agreeing to join the Army. However, the recruit does not become a member of the Army until he or she returns to the MEPS, is evaluated for continued eligibility, swears in to the active military and departs for a basic training installation.

During the period between signing the contract and the beginning of basic training, each recruit is enrolled in the DEP. While in DEP, the recruit is responsible to contact and verify eligibility with his or her recruiter on a regular

basis. It is the responsibility of recruiters to insure that each of their recruits eventually enters basic training. If, while in DEP, a recruit decides not to join the Army, i.e., he or she becomes a DEP loss, the recruiter must replace the DEP loss with a new contract of the same mission box classification.

III. OPTIMAL MISSIONING MODEL

As stated in the previous chapter, the PAE Directorate at USAREC is responsible for developing the quarterly recruiting or contract mission for each of the four recruiting brigades. To ensure that USAREC recruiting meets the annual requirements set by DA, PAE analysts currently set the contract mission higher than the accession mission to account for DEP losses. In effect, they assume that every contract has a 0.15 probability of becoming a DEP loss.

Although this practice is effective, USAREC seeks a more cost efficient way to account for DEP losses when planning their recruiting missions. In particular, the assumption that all contracts have a 0.15 probability of becoming a DEP loss is costly. In fact, different mission BOXES have different DEP loss probability. Thus, using a constant (0.15) DEP loss probability results in USAREC under missioning contracts for some mission BOXES and over missioning for cthers.

In this chapter, the constant DEP loss assumption is relaxed and each mission BOX and DEP length combination has a unique DEP loss probability. The estimation of these probabilities is discussed in the net chapter. For this chapter, the first section describes problem of planning recruiting missions (to be referred to as the missioning problem) and the second formulates it mathematically.

A. PLANNING RECRUITING MISSIONS

In planning recruiting missions for the quarter in question or missioning quarter, analysts of the Missioning Division must set the contract mission, i.e., determine the number of contracts needed in each category during each of the three months. The contract mission cannot be arbitrarily determined because it must be implementable, i.e., it must meet certain constraints imposed by DA requirements, operational necessities and future accession requirements.

1. DA Requirements

The DA requirements relate to the accession mission defined by the DCSPER. The DA mission includes volume, PS/NPS mix, quality and gender requirements. These requirements are discussed below.

a. Volume Requirement

This requirement insures that the total number of accessions satisfies the required number for each fiscal year (FY). For example, DA requires 75,000 accessions during both FY94 and FY95.

b. PS/NPS Mix Requirement

This requirement insures that a proper percentage of total accession have no prior military service (NPS). For example, DA requires that 93% of FY94's accessions and 96% of FY95's accessions have no prior military service.

c. Quality Requirement

This requirement insures that there is a proper quality mix among all NPS accessions. As stated in Chapter II, DCSPER requires that, among all NPS accessions for FY94, at least 95% must be high school graduates, at least 67% must score in the top 50th percentile of the AFQT and at most 2% can score between the 26th and 30th percentiles on the AFQT.

d. Gender Requirement

This requirement insures that at least 14.8% of the NPS accessions are female.

2. Operational Requirements

In general, these requirements insure that the contract mission is operationally feasible and yields an even workload for recruiters throughout the year.

a. Recruiter Workload

USAREC has approximately 4200 recruiters and each recruiter is expected to produce between one and two contracts per month. This places an upper limit on the number of contracts that can be produced each month.

b. Monthly Accession Limitation

The Training and Doctrine Command (TRADOC) is responsible for training the recruits inducted into the Army. The number of basic training and Advanced Individual Training (AIT) spaces, or training seats, varies from month to month to reflect the normal fluctuation in enlistment chroughout the

year. Typically, there are more training seats during the months following high school graduation dates and less during the major holidays. The second column in Table V lists TRADOC's programmed accessions during each month in FY94 and FY95.

MONTH	FY94 & FY95 ACCESSIONS	FY94 # OF WORKING DAYS	FY95 # OF WORKING DAYS
Oct	7666	19	24
Nov	8784	23	18
Dec	2059	19	19
Jan	5241	23	23
Feb	5959	19	19
Mar	4679	20	20
Apr	4400	20	20
May	5808	24	24
Jun	6354	20	20
Jul	7656	19	24
Aug	8015	25	20
Sep	<u>8397</u>	19	<u>19</u>
TOTAL	: 75018	250	250

Table V MONTHLY ACCESSION REQUIREMENT

c. Proportion of Annual Contracts Written Each Month

The number of contracts written in a month is a function of how many working days exist during the month. As illustrated by the last two columns in Table V, the number of working days varies depending on the number of holidays and weekends occurring during the month. To insure an even workload, the number of contracts written during each month should be proportional to the number of working days in that month. For example, since October has 19 of the total 250

contract days in FY94, then contracts written in October should be (approximately) 7.6% (19/250) of the total FY94 contracts.

d. Time in DEP Requirement

As mentioned before, every recruit enters the DEP once he or she signs the enlistment contract. The number of months a recruit remains in the DEP ranges from a minimum of 1 month to a maximum of 12 months. High school recruits must remain in the DEP until after their graduation date.

e. HSDG NPS-A Quality Mix Requirement

Operationally, it is more time efficient to recruit high school seniors than graduates. Recruiters request to meet with all seniors on a single visit to a high school. On the other hand, recruiters must generally visit high school graduates at their home or place of work individually. This requires more time and effort by the recruiter.

In terms of DEP loss, it is more effective to recruit high school graduates than seniors. Experience indicates that seniors exhibit higher DEP loss probabilities than do graduates with similar service, gender and mental aptitude.

To balance the time and DEP loss factors, USAREC requires that GMA make up approximately 70% of the combined GMA and SMA contracts. A similar rule applies to the GFA and

SFA mix. (See Table II in Chapter II for the description of these BOX classifications.)

3. Future Accession Requirements

Since recruiting is an unending process, USAREC typically requires recruiters to write contracts in the current FY that will access and, therefore, satisfy the accession requirement for the next FY. Without this requirement, recruiters would not have to write any contracts during the last few months of each fiscal year. For example, since there is a minimum of one month DEP, recruiters cannot write any contracts to access in the current year during September. Thus, USAREC typically imposes the following rules.

a. Volume Requirement

USAREC desires that 35% of the accessions for the next FY be contracted during the current FY. For example, this means that 26,250 (35% of 75,000) contracts written during FY94 should access in FY95.

b. Quality Requirement

USAREC desires that 45% of the NPS-A accessions for the next FY be contracted during the current FY. (See Table III in Chapter II for mission BOXES included in category NPS-A.) Recall that NPS accounts for 72,000 (96% of 75,000) of the FY95 requirement, and at least 67% of the NPS must possess an AFQT score category A. This means that 21,708 (45% of 67%

of 72,000) NPS-A contracts written in FY94 should access in FY95.

B. PROBLEM STATEMENT

The problem of setting the recruiting or contract mission is to determine the number of contracts to be written for each mission BOX and DEP length combination during each month of the missioning quarter. To be implementable, these contracts must satisfy the requirements in the previous section, be cost efficient and yield the least number of DEP losses.

C. PROBLEM FORMULATION

The problem of planning the contract mission (or the missioning problem) is formulated as a linear program. Implicit in this formulation is the assumption that a fractional number of contracts is acceptable. Considering the magnitude of the monthly quotas and realizing that the results are for planning purpose only, this is reasonable. The fractional number can be rounded to the nearest integer solution without significantly degrading the optimality of the solution. Furthermore, the planning horizon is divided into months and includes the first month of the missioning quarter up to September of the following FY. The inclusion of months beyond the missioning decisions on future missioning quarters.

In the formulation below, the contracts are assumed to be independent and the event of them becoming a DEP loss can be represented as a collection of Bernoulli random variables. Then, the number of accessions from a collection of N contracts in a given BOX, for a given DEP length and written in a given month is a (independent) Binomial random variable with parameters N and p, where p is the probability of accession for the collection. (Note (1-p) is the corresponding probability of DEP loss.) To make this model manageable, the number of accessions or DEP losses are stated in terms of their probabalistic expectations.

INDICES:

У	=	fiscal year, e.g., 1, 2
b	=	BOX classification, e.g., SMA,, HFB
a,	C =	months in the planning period, e.g., OCTFY1,, SEPFY1, OCTFY2,, SEPFY2
t	=	time spent in the DEP, e.g., 1,, 12
đ	=	category, e.g., PS, NPS, HSDG, NPS-A, TSC-4 and FEM
k	=	quarter, e.g., 1, 2,, 8
9:	The i repre	ndices "a" and "c" are alias indices. They both esent months in the planning period. In the

Note: The indices "a" and "c" are alias indices. They both represent months in the planning period. In the formulation below, "c" represents the month in which the contract was signed, and "a" represents the month the same individual accesses into the Army, i.e., begins basic training. Thus (a-c) represents the time an individual remains in the DEP (also denoted as t).

INDEX SETS:

 Ω_q = set of BOX classifications which belong to category q, e.g., Ω_{p_c} = (MPS and FPS), (See Table III in Chapter II.)

٨,	=	months	be	longing	to	FY	У
----	---	--------	----	---------	----	----	---

 θ_k = months belonging to quarter k

DATA:

- S_a^y = available training seats in month a of FY y
- f_{μ}^{y} = required proportion for category q in FY y
- d_{ba} = number of contracts in BOX b that were written prior to the mission guarter and expected to access during month a
- α₁₂ = proportion of FY2 volume requirement to be contracted during FY1
- β_{12} = proportion of FY2 quality requirement to be contracted during FY1
- w^y = number of monthly contracts to be produced by a recruiter during year y
- R_a = number of available recruiters in month a
- D_c = proportion of annual contracts that must be contracted during month c
- p_{bt} = probability of accession for BOX b with t months in DEP, (In the formulation below, t is represented as (a-c), i.e., the difference between accession and contract months.)
- N^y = number of accessions required for year y
- mp^k = proportion of GMAs among GMA and SMA contracts in quarter k
- fp^k = proportion of GFAs among GFA and SFA contracts in quarter k
- ff^k = proportion of GFAs and SFAs among female contracts with no prior service in quarter k

VARIABLE:

- X_{bca} = number of contracts to be recruited for BOX b in month c, and scheduled to access in month a.
- NOTE: To simplify the notation, X_{pca} is assumed to be valid for combinations of c and a which yield a time in DEP

of 1 to 12 months. All summations over indices c and a are also assumed to be over the range of valid combinations of c and a.

FORMULATION

The Missioning Problem

OBJECTIVE FUNCTION:

$$MINIMIZE \quad \sum_{b} \sum_{a} \sum_{c} (1 - p_{b, (a-c)}) * X_{bca}$$

CONSTRAINTS:

$$\sum_{b \in \Omega_{MPS}} \sum_{a \in \Lambda_y} (d_{ba} + \sum_c p_{b,(a-c)} * X_{bca}) \ge f_{NPS}^y * N^y \quad \forall y \quad (1)$$

$$\sum_{b \in \Omega_{g}} \sum_{a \in h_{y}} (d_{ba} + \sum_{c} P_{b, (a-c)} + X_{bca}) \geq$$

$$f_{q}^{y} + (\sum_{b \in \Omega_{gbc}} \sum_{a \in h_{y}} (d_{ba} + \sum_{c} P_{b, (a-c)} + X_{bca}))$$

$$\forall y \text{ and } q = HSDG, NPS-A, FEM$$

$$(2)$$

$$\sum_{b \in \Omega_{TBC-4}} \sum_{a \in A_{y}} (d_{ba} + \sum_{c} \mathcal{P}_{b, (a-c)} + X_{bca}) \leq$$

$$f_{TBC-4}^{y} + (\sum_{b \in \Omega_{RBP}} \sum_{a \in A_{y}} (d_{ba} + \sum_{c} \mathcal{P}_{b, (a-c)} + X_{bca})) \quad \forall y$$
(3)

$$\sum_{b} \sum_{c} \mathcal{P}_{b,(a-c)} * X_{bca} + \sum_{b} d_{ba} = S_{a}^{y} \qquad \forall y \text{ and } a \in \Lambda_{y} \quad (4)$$

.....

- ·

المراري المتدرية والممرد فيمرد المرد المتحدين التارين المراجع المرد المراجع والمراجع

.....

$$\sum_{b} \sum_{a} X_{bca} \leq w^{y} * R_{c} * D_{c} \qquad \forall \ y \text{ and } c \in \Lambda_{y}$$
(5)

$$\sum_{b} \sum_{a \in A_{2}} (d_{ba} + \sum_{c \in A_{1}} \mathcal{D}_{b, (a-c)} + X_{bca}) \ge \alpha_{12} + N^{2}$$
(6)

$$\sum_{b \in \Omega_{app-a}} \sum_{a \in \Lambda_2} (d_{ba} + \sum_{c \in \Lambda_1} P_{b, (a-c)} * X_{bca}) \geq$$

$$\beta_{12} * f_{MPS-A}^2 * f_{MPS}^2 * N^2$$
(7)

$$\sum_{c \in \Theta_k} \sum_{a} (X_{cNA, c, a}) = mp^k * \sum_{c \in \Theta_k} \sum_{a} (X_{cNA, c, a} + X_{SNA, c, a})$$
(8)
$$\forall k$$

$$\sum_{c \in \Theta_k} \sum_{a} (X_{GPA, c, a}) = fp^k + \sum_{c \in \Theta_k} \sum_{a} (X_{GPA, c, a} + X_{SPA, c, a})$$

$$\forall k \qquad (9)$$

$$\sum_{c \in \Theta_k} \sum_{a} (X_{aFA, c, a} + X_{SFA, c, a}) \ge ff^k * \sum_{b \in \Omega_{FBK}} \sum_{c \in \Theta_k} \sum_{a} X_{bca}$$
(10)
$$\forall k$$

$$\sum_{b \in \Omega_{rev}} \sum_{c \in \Theta_k} \sum_{a} X_{bca} \ge \frac{f_{rev}}{2} * \sum_{b \in \Omega_{rev}} \sum_{c \in \Theta_k} \sum_{a} X_{bca}$$
(11)
$$\forall y \text{ and } k$$
$$\sum_{b} \sum_{a} X_{b,c,a} = D_{c}^{y} * \sum_{b} \sum_{c \in A_{y}} \sum_{a} X_{b,c,a} \qquad \forall y \text{ and } c \in \Lambda_{y} \quad (12)$$

 $X_{bca} \ge 0 \quad \forall \ b, \ c, \ a \tag{13}$

The summand in the objective function is simply the expected number of DEP losses under the Binomial assumption. Summing over all possible valid combinations provides the total expected number of DEP losses from the missioning guarter to the end of the second FY.

Constraint (1) insures that there is a sufficient number of NPS contracts for each FY. The term inside the parenthesis is the expected number of accessions in month a. Summing over $b \in \Omega_{NPC}$ and $a \in \Lambda_y$ gives the number of NPS contracts for year y. The right hand side is simply the required minimum number of NPS accessions.

Constraints (2) and (3) guarantee that there are sufficient contracts in categories HSDG, NPS-A, FEM and TSC-4 to satisfy the required proportion. (See Table III in Chapter II for BOXES in these categories.) Similar to constraint (1), the summation on the left hand side is the expected number of contracts in category q, where q = HSDG, NPS-A, FEM and TSC-4. The summation on the right hand side represents the total number of NPS contracts. The ratio of the quantities on the left over the right hand side of the inequality gives the

proportion of contracts for category q. For q = HSDG, NPS-A and FEM this value must be greater than or equal to the required proportion, f_q^{y} . For q = TSC-4 the ratio can not exceed the required proportion, f_{TSC-4}^{y} .

Constraint (4) insures that the number of accessions equals the programmed basic training seats for each month. The summations on the left hand side represent the expected number of recruits to access during month a. The first summation represents accessions due to contracts written during or after the missioning quarter, and the second represents accessions due to contracts written before the missioning quarter. The right hand side is the number of basic training seats available in month a.

Constraint (5) places an upper limit on the number of contracts that can be written each month. The summation on the left hand side is the number of contracts written during month c. On the right hand side, w^y , R_c and D_c represent the maximum number of contracts that a recruiter can write each month during year y, the number of available recruiters during month c and the proportion of working days in month c that are available for recruiting, respectively. The product, $w^y * R_c * D_c$, represents the maximum number of contracts that can be written during month c.

Constraint (6) guarantees that the required proportion of FY2's accessions are contracted during FY1. The summation on the left hand side is the expected number of contracts written

in FY1, i.e., $c \in \Lambda_1$, that will access in FY2, i.e., $a \in \Lambda_2$. This sum must be at least equal to the required number $\alpha_{12}N^2$, where α_{12} represents the minimum proportion of FY2's accessions that must be contracted for during FY1, and N² is the total number of accessions required for FY2. Similarly, constraint (7) insures that the required proportion of FY2's accessions in the NPS-A category are contracted during FY1.

Constraint (8) guarantees that the proportion of GMA contracts among GMA and SMA contracts meets the desired proportion, mp^k, for each quarter. Constraint (9) applies to GFA and SFA in the same manner as constraint (8). (Recall from Table II in Chapter II that G, S, M, F and A represent graduate, senior, male, female and an AFQT score in the top 50th percentile, respectively.)

Constraint (10) insures that there are a desired proportion, ff^k , of GFA and SFA contracts among contracts in category FEM in each quarter. In the same manner, constraint (11) applies to contracts in category FEM with the desired proportion $F_{FEM}^{y}/2$.

Finally, constraint (12) insures that the proportion of contracts written in each month is the same proportion of working days in that month over the total number of working days of each year in the planning horizon. Constraint (13) simply insures that the number of contracts is nonnegative.

One of the most important input data for the missioning problem is the probability of DEP loss, $(1-p_{bt})$. This

probability determines the additional number of contracts USAREC must produce in order to offset the losses during DEP. The estimation of this probability is the topic of the next chapter.

IV. ESTIMATING DEP LOSS PROBABILITIES

The DEP loss probability is the probability that a recruit becomes a DEP loss prior to leaving the MEPS for basic training or AIT. Intuitively, this probability varies with the time a recruit must stay in the DBF. From past experience, USAREC analysts indicate that the DEP loss probability is also a function of gender, mental capability. education, age, military service history, local unemployment rate, career choice (i.e., Military Occupational Specialty or MOS), etc. To limit the scopy of this thesis, the sections below focus on estimating DEP loss probability as a function of time in DEP and the 22 mission BOX classifications. In doing so, factors such as gender, education, military service history and mental capability are included in the estimation. In addition, such estimation is well suited for the missioning problem described previoualy.

The outline for this chapter is as follows. The first section describes the data base maintained by USAREC. The second section discusses some preliminary analysis and irregularities in the data due to changes in recruiting policy during the past five years. The third section tests the overall differences in DEP loss probabilities by quarter. Finally, the last section provides procedures for estimating the DEP loss probabilities.

A. DESCRIPTION OF DATA

.

USAREC maintains records of all recruits that have been processed at the MEPS in files generally referred to as the MINIMASTER files. These files contain information concerning the recruits such as their social security number, responsible recruiting station, age, etc. Among these, contract date, accession date, mission BOX designation, AFQT score and contract status (i.e., in DEP, accessed or DEP loss) are pertinent to this thesis.

Note that the difference between contract and accession dates provides the duration each recruit agreed to spend in the DEP. Table VI below summarizes the records or contracts in the MINIMASTER files from FY88 to FY92. Note that not all contracts are considered; reasons for their deletion are indicated therein. In particular, mission BOX 23 designates special forces recruits, which are not considered in this thesis. Table VII lists the number of contracts by year and quarter.

Table VI MINIMASTER FILES FROM FYP? 1/92

Total Records (Contracts) Number of Records Deleted		= 535,746 = 6,213
BOX 23 (Special Forces) Missing Dates Wrong Record Type FY92 Open Records	= 1,996 = 163 = 1 = 4,053	- 0,210
TOTAL records remaining	-	= 529,533

Table VII	NUMBER	OF RECORI	DS BY YEAR	R AND QUAR	TER
		OUAR	TER		
YEAR	1	2	3	4	TOTAL
FY88	29,497	33,610	28,656	33 552	125,315
FY89	33,610	35,050	33,703	33 931	136,294
FY90	32,385	30,269	18,781	23 171	104,606
FY91	24,736	21,432	19,933	24,653	90,754
FY92	18,315	19,492	16,201	18,556	72,564
TOTAL	138,543	139,853	117,274	133,863	529,533

B. PRELIMINARY ANALYSIS

Figure 2 displays the probability of DEP loss by quarter and year as computed from 529,533 contracts produced during FY88 to FY92.



Figure 2 Preliminary DEP Loss Probabilities

From this figure, the DEP loss probabilities from quarters 1, 2 and 3 of FY90 and quarters 1, 2 and 4 of FY92 appear to be different than those from other fiscal years. To test whether these differences are statistically significant, hypothesis tests are conducted for the six quarters listed above. To illustrate, the hypotheses for quarter 1 of FY90 are:

£ .

 $H_o: p_1^{90} = p_1^{POOLED}$ (Null Hypothesis)

 H_a : $p_1^{90} \neq p_1^{POOLED}$ (Alternate Hypothesis)

where p_1^{90} is the DEP loss probability from quarter 1 of FY90 and p_1^{POOLED} is the DEP loss probability from quarter 1 pooled over FY88, FY89 and FY91. The following Chi-Squared statistic [Ref. 3] is used to perform the test:

$$\chi^2 = \sum_{i} \frac{(O_i - \Theta_i)^2}{\Theta_i}$$

where o_i and e_i , i = 1, ..., 4, represent the observed and expected number of observations for the cells in Table VIII. The expected frequencies are calculated as

$$e_{i} = \frac{(column \ total) \times (row \ total)}{total \ contracts}.$$

Using the data from Table VIII, the Chi-Squared statistic is computed as

$$\chi^{2} = \frac{(27653 - 28572)^{2}}{28572} + \frac{(78420 - 77501)^{2}}{77501} + \frac{(4732 - 3813)^{2}}{3813} + \frac{(9423 - 10342)^{2}}{10342} = 344.$$

	<u>YEA</u> FY90	ROW <u>TOTAL</u>	
ACCESSIONS DEP LOSS	27,653(28,572)* <u>4,732</u> (3,813)*	78,420(77,501)* <u>9,423</u> (10,342)*	106,073 <u>14,155</u>
COLUMN TOTAL	32,385	87,843	120,228
& DEP LOSS	14.61	10.73	11.77
* expected	frequencies		

Table VIII EXAMPLE DATA NEEDED FOR CHI-SQUARED TEST

Since the calculated Chi-Squared statistic (344) is greater than the $\chi^2_{\alpha,0.05}$ with 1 degree of freedom (3.841), the null hypothesis is rejected. So, the DEP loss probability associated with quarter 1 of 1990 is significantly different from the probability pooled over FY88, FY89 and FY91. Using the same Chi-Squared test, the DEF loss probabilities from each of the remaining five quarters (i.e., quarters 2 and 3 of FY90 and quarters 1, 2 and 4 of FY92) are also significantly different from the pooled DEP loss probabilities with pvalues less than .05 (see the complete set of results in Appendix A). The fact that the data gives statistical significance is not too surprising given the large sample sizes used for each test. However, the differences in DEP loss probabilities also have practical significance.

When the above differences in DEP loss probabilities were conveyed to a senior USAREC analyst, he explained that during the early 90's there were changes in recruiting policy which resulted in "DEP purges." These purges encouraged recruiters to drop recruits from the DEP pool in order to achieve the lowered accession mission. In guarter 4 of FY92, the situation was reversed. As a result of DEP purges during earlier quarters, USAREC was in danger of not meeting the accession mission for the year. This induced an extra burden on the recruiters to prevent DEP loss at all costs, thereby producing an unusually low DEP loss probability for the last quarter of FY92. Because of these unusual circumstances, the records from the effected quarters are removed and the remaining data set reduces to 391,735 records. Table IX provides a summary of the reduced data set.

<u>YEAR</u> FY88 FY89 FY90 FY91 FY92	<pre># OF OBSERVATIONS 125,315 136,294 23,171 90,754 16,201 </pre>
TOTAL	391,735
OUARTER 1 2 3 4	# OF OBSERVATIONS 87,843 90,092 98,493 <u>115,307</u>
TOTAL	391,735

Table IX OBSERVATIONS USED FOR ANALYSIS

C. DEP LOSS PROBABILITY BY QUARTER

USAREC analysts indicate that they experience a higher percentage of DEP loss during the summer months (quarter 4). During these months, DEP recruits are out of school and much of their time is uncommitted. This enables them to pursue more career options, to be influenced by others and, in unfortunate cases, to be involved in an accident causing them to become ineligible for military service. To test this observation, the following hypotheses are considered:

 $H_0: p_1 = p_2 = p_3 = p_4$

 H_a : p_i are not equal

where p_i = the DEP loss probability for quarter i.

Using the same procedure described in the previous section and the data in Table X, the Chi-Squared statistic evaluates to 675.385.

	1	2 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	TER 3	4	TOTAL
ACCESSIONS DEP LOSS	78,420 <u>9,423</u>	81,339 <u>8,753</u>	87,016 <u>11,477</u>	100,054 <u>15,253</u>	346,829 <u>44,906</u>
TOTAL	87,843	90,092	98,493	115,307	391,735
& DEP LOSS	10.73	9.72	11.65	13.23	11.46

Table X QUARTERLY ACCESSIONS AND DEP LOSS

Since the $\chi^2_{\alpha=0.05}$ with 3 degrees of freedom is only 7.815, the null hypothesis is again rejected (see Appendix A for further details).

So, the DEP loss probabilities significantly differ from quarter to quarter. These differences can be seen in Figure 3. Note that the DEP loss probability for quarter 4 is the largest which seems to corroborate the experience of USAREC analysts.



Figure 3 Quarterly DEP Loss Probability

However, the above test does not conclusively state that each probability, p_i , differs from the others. To test whether any two probabilities are the same, six pairs of hypotheses of the following form

- $H_o: p_i = p_j$
- H_a : $p_i \neq p_j$

are tested. In all cases, the null hypothesis is rejected (see Appendix A for further details), thereby implying that the DEP loss probabilities for each quarter are different and should be computed separately.

D. ESTIMATING DEP LOSS PROBABILITIES

One approach for estimating the probabilities is to view the number of DEP losses for a given BOX and DEP duration as a binomial random variable with parameters N_{bt} and p_{bt} . Here, N_{bt} is the number of contracts in BOX b with DEP duration t, and p_{bt} is the corresponding probability of DEP loss. Then, an unbiased estimator for p_{bt} is given by [Ref. 3]

$$\stackrel{\wedge}{P}_{bt} = \frac{L_{bt}}{N_{bt}}$$

where L_{br} = the number of DEP losses in BOX b with DEP duration t. This estimate is used for BOX and DEP length combinations which contain at least 30 observations or contracts. Assuming normality, one can be $(1-\alpha)100$ % confident that the error in the above estimate will not exceed [Ref. 3]

$$Z_{a/2}\sqrt{\frac{\begin{array}{c} A \\ p \\ bt \end{array}}{\frac{bt}{N_{bt}}}}.$$

To illustrate, Figure 4 displays the probability of DEP loss for GMA, along with the 95% confidence interval.



Figure 4 First Quarter DEP Loss For GMA

'The other BOXES display similar shape (see Appendix B). In general, DEP loss probabilities tend to increase as the DEP length increases. The "dips" in DEP loss probability after months seven and eleven are counter-intuitive. One USAREC analyst indicated that recruits with a twelve month DEP length are generally assigned the MOS of their choice and are less likely to become a DEP loss.

Among the 22 mission BOXES, ten contain fewer than 30 observations for some DEP durations. These include two BOXES (SM4 and SF4) that are not being recruited, and eight BOXES (SFB, GF4, NMB, NFA, NFB, HMB, HFA and HFB) which have low priority. For each of the eight BOXES, it is assumed that the probability of DEP loss is the same for all four quarters. This assumption is valid for BOXES that do not involve high school seniors. The assumption remains valid for SFB contracts since almost every one of them became a DEP loss, regardless of when they were contracted. Under this assumption, isotonic regression [Ref. 4] is used to estimate the DEP loss probabilities. In isotonic regression, an estimate of DEP loss probability for a given BOB b, \hat{p}_{bt} , is taken to be an optimal solution to the following problem:

$$MIN\sum_{t} W_{bt} * (f_{bt} - p_{bt})^2$$

subject to

$$p_{bt} \leq p_{b,t+1} \quad \forall \ t < 11 \tag{14}$$

$$p_{b, 11} \ge p_{b, 12}$$
 (15)

$$P_{b,1} \leq P_{b,12}$$
 (16)

$$0 \leq p_{ht} \perp 1 \quad \forall t \tag{17}$$

where w_{bt} represents a weight for DEP duration t and is taken to be the number of observations and f_{bt} is the observed probability of DEP loss.

Recall that the DEP loss probabilities fluctuate in a counter-intuitive manner. Based on discussions with USAREC analysts, the estimated DEP loss probabilities should be nondecreasing in t up to month eleven and nonincreasing thereafter. Constraints (14) and (15) insure these two

assumptions are satisfied. Constraint (16) insures that a twelve month DEP loss probability estimate is no smaller than that for a one month DEP duration. Constraint (17) insures that the probability estimates are nonnegative. As an example, Table XI compares the isotonic regression estimates for GF4 against the observed DEP loss probabilities (f_{bt}) most of which are based on less than 30 observations (w_{bt}) . These estimates are obtained by solving the above optimization problem using the General Algebraic Modeling System or GAMS [Ref. 5]. Results from other low priority BOXES are in Appendix B.

DEP	W _{bt}	f _{bt}	$\mathbf{\hat{p}}_{\mathrm{ist}}$
$\frac{DORATION}{1}$	60	0.067	0.106
2	8	0.375	0.419
3	4	0.250	0.419
4	5	0.400	0.419
5	1	0.000	0.419
6	0	0.000	1.000
7	0	0.000	1.000
8	0	0.000	1.000
9	1	1.000	1.000
10	0	0.000	1.000
11	0	0.000	1.000
12	5	0.000	0.106

Table XI RESULTS FROM ISOTONIC REGRESSION FOR GF4

V. MODEL IMPLEMENTATION AND ANALYSIS

The missioning problem described in Chapter III was implemented using GAMS [Ref. 5] on a 80486-33Mhz personal To insure that the model always produces a computer. solution, all constraints are made elastic. To do so, an artificial (or elastic) variable is added to each constraint to represent the amount of possible violation. In the objective function, each artificial variable is assigned a When the missioning problem has a large penalty cost. feasible solution, the optimal values for these artificial variables are zero. Otherwise, an optimal solution with nonzero artificial variables indicates the problem is infeasible and provides information as to which constraints can not be satisfied. To make the problem feasible, USAREC analysts can then modify their input data or parameters accordingly. The sections below describe data inputs, model outputs and sample analyses.

A. DATA INPUTS

Data necessary for determining an optimal contract mission for the first quarter of FY94 is used to illustrate and validate outputs from the model. The basic data inputs are summarized in Table XII. The number of required accessions and working days in each month are the same as those listed in

Table V of Chapter III. The remaining inputs are given in Appendices C and D and include the DEP loss probability matrix and the number of accessions generated by contracts from previous quarters, respectively.

PARAMETER	FISCAL <u>FY94</u>	YEAR FY95
ACCESSIONS (Volume) SERVICE MIX	75000	75000
NPS	70000	72000
PS	5000	3000
QUALITY (NPS)		
High School Graduate	95.0%	95.0%
AFQT Category A	67.0%	67.0%
AFQT Category 4	2.0%	2.0%
FEMALE (NPS)	14.8%	14.8%
FY95 VOLUME CONTRACTED		
DURING FY94	N/A	35.0%
FY95 QUALITY CONTRACTED		
DURING FY94	N/A	45.0%
# OF OPRA RECRUITERS	4200	4200
CONTRACTS/RECRUITER/MONTH	*2.0	1.9
* 1.9 is infeasible		

Table XII USER DEFINED PARAMETER VALUES

It is important that USAREC meets the annual recruiting noission set forth by DCSPER. Thus, to insure that the solution has a high probability of meeting the DCSPER mission, the probability of DEP loss in Appendix C is actually the upper limit of the 97.5% one sided confidence interval. The upper limit is given by

$$\bigwedge_{\substack{p\\bt}}^{\wedge} + 1.96 \neq \sqrt{\frac{\bigwedge_{p}^{\wedge} (1 - \bigwedge_{p}^{\wedge})}{\frac{bt}{N_{bt}}}}$$

where, as before, p_{bt} and N_{bt} are the point estimate of the DEP loss probability and the number of observations for BOX b and DEP duration t, respectively. The number 1.96 is the 97.5 quantile of the standard normal distribution. When isotonic regression is used to estimate the DEP loss probability, N_{br} is set at 30 to insure that the upper limit is no larger than one.

Part of the inputs to the missioning model are the estimates of how many contracts (by mission BOX) from previous quarters will actually access into the Army during the missioning quarter and beyond. These estimates should be functions of the conditional DEP loss probability which is a topic for future research. In this implementation, USAREC analysts provided the estimates which they refer to as the NETDEP matrix (see Appendix D).

B. MODEL OUTPUTS

Using the input described above, the resulting optimization problem contains 3,899 (continuous) variables and 115 constraints. The problem requires 1.8 cpu minutes on the 80486-53Mhz personal computer to produce an optimal solution.

The GAMS implementation of the missioning model summarizes the optimal solution in the form of output reports. There are a total of five different reports (see Appendix E). Four of which provide information regarding the feasibility of the

solution obtained by GAMS. The remaining report provides an optimal contract mission and is discussed below.

Critical Missioning Report: This report provides the optimal contract mission for each BOX during FY94 and FY95. However, only the contract mission for the first quarter of FY94 is of interest at this time. Information for other quarters should be considered as notional and is provided for long-range planning purposes. Table XIII below lists the recommended contract mission for the first quarter of FY94.

	<u>OCTFY1</u>	NOVFY1	DECFY1
SMA		4063	
GMA	1146	1951	6384
GM4		1186	
NMA	2648	156	
SFA	250	372	
GFA	1452		
GFB	888		
CONTRACTS EXPECTED	6384	7728	6384
ACCESSIONS	5871	6812	5958
RECRUITER CAPACITY SHORTFALL	6384	7728	6384

rable X	<u>lili</u>	RECOMMENDED	CONTRACT	MISSION FOR	QTR1,	FY94
---------	-------------	-------------	----------	-------------	-------	------

The first row of Table XIII indicates that in November of FY94 USAREC should produce 4063 contracts which are high school seniors, male and have AFQT scores in the upper 50th percentile. Except for the last four rows, the other rows provide similar information. Three of the last four rows are the number of contracts written, accession expected and recruiting capacity for each month. The recruiting capacity is the number of contracts that can be written in a given This number is based on the number of working or month. recruiting days in a given month (see Table V in Chapter III), the number of recruiters and the write-rate of 2.0 contracts per month. If the number of contracts exceed the recruiting capacity in a given month, SHORTFALL will contain a positive number representing the number of contracts over the capacity, indicating that the inputs yield an infeasible problem. In this case, the analyst must increase the number of recruiters, write-rate or reduce the number of accessions required during a given month. In addition, the difference between the number of expected accessions and contracts is the number of DEP losses resulting from each month's contracts. For example, of the 6384 contracts written in October, 513 are expected to become DEP losses.

C. SAMPLE ANALYSES

To illustrate the potential of the missioning problem as a tool to aid in the decision making, two sample analyses are provided below. One concerns the proportion of females and the other concerns the contracts produced in FY94 to satisfy requirements in FY95.

1. **Female** Proportion

Using the data described in chapter II, one can estimate the DEP loss probabilities for each sex. Figure 5 below displays the resulting calculations graphically.

.



It is clear from Figure 5 that the DEP loss probabilities for females are generally higher than males. Thus it is of concern as to how the required proportion of females effects the cost of recruiting as measured by the number of DEP losses. Table XIV summarizes the solutions from the missioning problem using various values for the required female proportion in the DCSPER mission.

REQUIRED	FY94		
% NPS	EXPECTED %	EXPECTED # OF	COST OF
FEMALES	DEP LOSS	DEP LOSSES	DEP LOSSES
10.8	8.40	7053	\$35,265,000
12.8	8.49	7134	\$35,670,000
14.8	8.59	7216	\$36,080,000
16.8	8.69	7297	\$36,485,000
18.8	8.77	7366	\$36,830,000
20.8	8.85	7437	\$37,185,000

Table XIV NPS FEMALE SENSITIVITY ANALYSIS

In the last column, the cost of DEP losses is based on the average cost of \$5000 to produce one contract [Ref. 2]. In any case, the results from this table suggest that when the mission is optimally determined, an increase in female proportion does not significantly increase the cost of recruiting. In fact, a 93% increase in proportion from 10.8% to 20.8% only results in a 5.44% increase in the number of DEP losses. It should be noted that these results are due in part to the assumption that the supply of contracts is unlimited for all BOX and DEP length combinations.

2. Contracts Produced in FY94 for FY95 Accession

To insure a smooth transition in recruiting from one FY to another, USAREC requires that a proportion of the accessions for the next FY is contracted during the current FY. This practice forces recruits to remain in DEP longer, thereby increasing their chance of becoming a DEP loss. To analyze the effect of this policy, consider the current

practice of recruiting 45% of FY95's NPS-A quality contracts during FY94. As before, Table XV below summarizes the results from the missioning problem as proportion is varied from

35% to 55%.

able XV FY95	NPS-A ACC	JNIRACTE	ED DURING FY94
% OF FY95'S			
ACCESSIONS	FY94	EXPECTED	
CONTRACTED	EXPECTED %	# OF	COST OF
DURING FY94	DEP LOSS	DEP LOSSES	DEP_LOSSES
35.0	8.38	7039	\$35,195,000
40.0	8.48	7125	\$35,625,000
45.0	8.59	7216	\$36,080,000
50.0	8.74	7345	\$36,725,000
55.0	*8.86	*7444	\$37,220,000
55.0	*8.86	*7444	\$37,220

Table XV shows that the current number of recruiters (4200) is insufficient to recruit 55% of FY95's NPS-A accessions during FY94. However, the 43% increase in NPS-A accessions from 35% to 50% results in a 4.35% increase in DEP losses. This increase is more significant than the one due to the increase in female proportion.

VI. CONCLUSION

A. SUDGARY

This thesis addresses the problem of determining the number of contracts to be recruited each month by Army recruiters. This missioning problem involves two areas of operations research: statistics and optimization. The statistical aspect involves estimating the probabilities that contracts will eventually become accessions based on the binomial assumption and isotonic regression. Using the probability estimates, the problem is formulated as a linear program that maximizes the expected number of accessions (or minimizes the number of DEP losses). The constraints in the problem include requirements established by DCSPER and USAREC.

The model for the missioning problem is implemented in GAMS and tested using data provided by USAREC analysts. Model outputs were presented and validated by USAREC analysts. Among these outputs is the optimal monthly level of contract mission for each mission BOX during the current and following fiscal years.

Although USAREC assumes a 15.0% DEP loss when setting their monthly mission, Table X in Chapter IV shows that only 11.46% of their contracts became DEP losses. Moreover, the model outputs indicate that, when the contracts are optimally

missioned, the expected DEP loss probability can be reduced from 11.46% to 8.59%. This translates to nearly \$11 million annual saving, a considerable sum during an austere budget environment.

د مهمور مدر مارد الألال في مناصر في الا ألال الا

In addition to providing the optimal missioning levels, the model can be used as a tool to analyze the impact of various policies. In particular, Chapter V examines two policies. One concerns the percentage of females among the new recruits to be accessed into the Army. Historically, females have higher DEP loss probability than males. The model is used to compute the expected DEP loss and associated cost as the percentage of required females varies between 10.8% and 20.8%. The other policy concerns the percentage of NPS-A quality recruits. Again, the model is used to evaluate the impact in terms of DEP loss and associated cost as the percentage of required NPS-A quality recruits is increased.

B. AREAS FOR FUTURE RESEARCH

This thesis identifies two topics for future research. They are described below.

1. Conditional DEP Loss Probabilities

As discussed in Chapter V, part of the inputs to the missioning model requires users to estimate the number of accessions that would be generated by contracts written during previous quarters. Such estimates should be functions of the conditional DEP loss probability. Assume that an individual

is required to be in DEP for ten months. Given that eight months have elapsed and the individual has not become a DEP loss, the DEP loss probability should be conditioned on the fact that he or she has already been in DEP for eight months. Statistical techniques that accurately estimate the conditional DEP loss probabilities must take into account the fact that recruiters often delay reporting DEP losses in order to make the recruiting mission in a given month.

2. Missioning by Brigade

The model described in Chapter III addresses the missioning problem at the headquarters level. The next step in the process is to allocate the monthly mission at the headquarters level to the next level in the hierarchy, which is the brigade. Such allocations should be optimal, e.g., minimize expected DEP losses and consider the socio-economic factors, such as unemployment rate, education level and median income, in each brigade's recruiting territory. These socioeconomic factors indirectly determine the supply of individuals with high propensities to enlist into the military in each of the geographical areas.

APPENDIX & CHI-SQUARED TEST

A. SUMMARY

.

Ho	H.	X²	p-value	Decision
$\mathbf{p}_1^{90} = \mathbf{p}_1^{\text{PUDLED}}$	$P_1^{90} \neq P_1^{POOLED}$	343.745	<.00005	Reject H _o
$\mathbf{D}_{2}^{90} = \mathbf{D}_{2}^{POOLED}$	$P_2^{90} \neq P_2^{POOLED}$	1005.196	<.00005	Reject H.
$D_1^{90} = D_1^{PODLED}$	$D_1^{90} \neq D_1^{POOLED}$	348.675	<.00005	Reject H.
$D_{1}^{92} = D_{1}^{POOLED}$	$D_1^{92} \neq D_1^{POOLED}$	263.041	<.00005	Reject H
$D_2^{92} = D_2^{POULED}$	$D_2^{92} \neq D_2^{POOLED}$	202.219	<.00005	Reject H
$p_4^{92} = p_4^{\text{POOLED}}$	$p_4^{92} \neq p_4^{\text{POOLED}}$	112.432	<.00005	Reject H _o
$\mathbf{p}_{01} = \mathbf{p}_{02} =$	At least			
$p_{03} = p_{04}$	one pair ≠	675.385	<.00005	Reject H
$\mathbf{p}_{\rm ul} = \mathbf{p}_{\rm u2}$	$\mathbf{p}_{01} \neq \mathbf{p}_{02}$	49.613	<.00005	Reject H
$\mathbf{p}_{\mathrm{o}1} = \mathbf{p}_{\mathrm{o}1}$	$\mathbf{p}_{01} \neq \mathbf{p}_{01}$	39.939	<.00005	Reject H
$\mathbf{D}_{c,1} = \mathbf{D}_{c,4}$	$\mathbf{p}_{c1} \neq \mathbf{p}_{c4}$	292.269	<.00005	Reject H.
$\mathbf{D}_{c,2} = \mathbf{D}_{c,3}$	$\mathbf{p}_{02} \neq \mathbf{p}_{03}$	184.343	<.00005	Reject H
$\mathbf{D}_{12} = \mathbf{D}_{14}$	$\mathbf{D}_{02} \neq \mathbf{D}_{04}$	604.565	<.00005	Reject H
$P_{03} = P_{04}$	$\mathbf{P}_{03} \neq \mathbf{P}_{04}$	120.541	<.00005	Reject H
	H_{0} $p_{1}^{90} = p_{1}^{900LED}$ $p_{2}^{90} = p_{2}^{900LED}$ $p_{3}^{90} = p_{3}^{900LED}$ $p_{1}^{92} = p_{1}^{900LED}$ $p_{2}^{92} = p_{2}^{900LED}$ $p_{4}^{92} = p_{4}^{900LED}$ $p_{4}^{91} = p_{02} =$ $p_{01}^{91} = p_{02}$ $p_{01}^{91} = p_{03}$ $p_{01}^{91} = p_{04}$ $p_{02}^{92} = p_{04}$ $p_{02}^{92} = p_{04}$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

Chi-Squared Test

 $\begin{array}{rcl} H_{o} & : & p_{1}^{90} = p_{1}^{POOLED} \\ H_{a} & : & p_{1}^{90} \neq p_{1}^{POOLED} \end{array}$

POOLED = Pooled average of guarter 1 from 1988, 1989 and 1991

	YEAR			
BECORD		POOLED	1990	Row Total
ACCESS	3	78420 	27653	106073 88.2
DEP LOSS	4	9423 	47321	14155 11.8
		++		
	Column Total	87843 73.1	32385 26.9	120228 100.0

Chi-Square	Value	DF	Significance
Pearson	343.74538	1	.00000
Likelihood Ratio Mantel-Haenszel test for linear	330.63119	ī	.00000
association	343.74252	1	.00000

The $\chi^{2}_{\alpha=0.05, v=1}$ value is 3.841.

Since 343.745 > 3.841, reject H.

$$\begin{array}{rcl} H_{\circ} & : & p_2^{90} = p_2^{\text{POOLED}} \\ H_{\bullet} & : & p_2^{90} \neq p_2^{\text{POOLED}} \end{array}$$

•

POOLED = Pooled average of guarter 2 from 1988, 1989 and 1991

	YEAR				
RECORD		POOLED	1990	Row Total	
ACCESS	3	81339 	253021	106641 88.6	
DEP LOSS	4	1 8753 	4967	13720 11.4	
	Column Total	90092 74.9	30269 25.1	120361 100.0	

Chi-Square	Value	DF	Significance
Pearson	1005.19591	1	.00000
Likelihood Ratio Mantel-Haenszel	937.99627	1	.00000
linear association	1005.18755	1	.00000

The $\chi^2_{\alpha=0.05, v=1}$ value is 3.841.

Since 1005.196 > 3.841, reject H_o .

 $\begin{array}{rcl} H_{o} & : & p_{3}^{90} = p_{3}^{\text{PODLED}} \\ H_{a} & : & p_{3}^{90} \neq p_{3}^{\text{PODLED}} \end{array}$

POOLED = Pooled average of quarter 3 from 1988, 1989, 1991 and 1992

	YEAR				
RECORD		POOLED	1990	Row Total	
ACCESS	3	87016 	15671	102687 87.6	
DEP LOSS	4	11477 	3110	14587 12.4	
	Column Total	98493 84.0	18781 16.0	117274 100.0	

Chi-Square	Value	DF	Significance
Pearson	348.67477	1	.00000
Likelihood Ratio Mantel-Haenszel test for linear	325.91111	1	.00000
association	348.67180	1	.00000

The $\chi^{2}_{\alpha=0.05, v=1}$ value is 3.841.

Since 348.675 > 3.841, reject H_{o} .

· · · · · · ·

 $\begin{array}{rcl} H_{o} & : & p_{1}^{92} = p_{1}^{POOLED} \\ H_{o} & : & p_{1}^{92} \neq p_{1}^{POOLED} \end{array}$

POOLED = Pooled average of quarter 1 from 1988, 1989 and 1991

	YEAR			
RECORD		POOLED	1992	Row Total
ACCESS	3	784201	15582 	94002 88.5
DEP LOSS	4	9423 	2733 	12156 11.8
	Column	87843	18315	106158

10001 02.7 17.5 100.0	Total	82.7	17.3	100.0
-----------------------	-------	------	------	-------

Chi-Square	Value	DF	Signific

Pearson	263.04141	1	.0000(
Likelihood Ratio Mantel-Haenszel test for linear	247.26244	1	.00000
association	263.03894	1	.00000

The $\chi^2_{\alpha=0.05, v=1}$ value is 3.841.

Since 263.041 > 3.841, reject H_o.

 $\begin{array}{rcl} H_{o} & : & p_{2}^{92} = p_{2}^{\text{Projled}} \\ H_{a} & : & p_{2}^{92} \neq p_{2}^{\text{Projled}} \end{array}$

POOLED = Pooled average of quarter 1 from 1988, 1989 and 1991

	YEAR			
RECORD		POOLED	1992	Row Total
ACCESS	3	813391	169321	98271 89.7
DEP LOSS	4	8753 	2560 	11313 10.3
	Column	5.3092	19492	109584

COTMIN	20022	13474	103204
Total	82.2	17.8	100.0

Chi-Square	Value	DF 	Significance
Pearson	202.21945	1	.00000
Likelihood Ratio Mantel-Haenszel test for	191.10495	ī	.00000
association	202.21760	1	.00000

The $\chi^{2}_{\alpha=0.05, v=1}$ value is 3.841.

i

•

.

Since 202.219 > 3.841, reject H_o.

 $\begin{array}{rcl} H_o & : & p_4^{92} = p_4^{PODLED} \\ H_a & : & p_4^{92} \neq p_4^{PODLED} \end{array}$

POOLED = Pooled average of quarter 4 from 1988, 1989, 1990 and 1991

	YEAR				
PECOPD		POOLED	1992	Row Total	
RECORD	3	11000541	166221	116676	
ACCESS		1 1		87.2	
DEP LOSS	. 4	15253 	+ 1934 	17187 12.8	
	++				
	Column	115307	18556	133863	

orumn	112301	18220	T33803
Total	86.1	13.9	100.0

Chi-Square	Value	DF	Significance
Pearson	112.43177	1	.00000
Likelihood Ratio Mantel-Haenszel test for linear	118.04553	1	.00000
association	112.43093	1	.00000

The $\chi^2_{\alpha \star 0.05,\ v\star 1}$ value is 3.841.

Since 112.432 > 3.841, reject H_o .

		I QUARTER			
RECORD		 1	21	31	Row 4 Total
ACCESS	3	784201	81339 	87016 1000	54 346829 88.5
DEP LOSS	4	9423 	8753 	11477 152 	253 44906 11.5
	Column Total	87843 9 22.4	0092 9 23.0	8493 115307 25.1 29.4	7 391735 1 100.0

Chi-Square	Value	DF	Significance
Pearson	675.38546	3	.00000
Likelihood Ratio Mantel-Haenszel test for linear	674.62498	3	.00000
association	474.69741	1	.00000

The $\chi^2_{\alpha=0.05,\ v=3}$ value is 7.815.

Since 675.385 > 7.815, reject H_o.



The $\chi^{2}_{\alpha=0.05, v=1}$ value is 3.814.

Since 49.613 > 3.814, reject H_o.

association 49.61277 1

.00000
ı.

C	count l	QUAR	TER	
	1	11	31	Row Total
RECORD	3	78420	87016 	165436 88.8
DEP LOSS	4	94231	11477	20900 11.2
	Column Total	87843 47.1	98493 52.9	186336 100.0

Chi-Square	Value	DF 	Significance
Pearson	39.93900	1	.00000
Continuity			
Correction	39.84611	1	.00000
Likelihood			
Ratio	40.00410	1	.00000
Mantel-Haenszel test for			
linear			
association	39.93879	1	.00000

The $\chi^2_{\alpha=0.05, v=1}$ value is 3.814.

Since 39.939 > 3.814, reject H_o .

Chi-Squared Test

या जाता विविधानी परिवर्त प्राप्त के विविधानी देवी के प्रतिहर्त के प्राप्त के प्राप्त के प्राप्त के प्

 $\begin{array}{ll} H_o: & p_{Q1} = p_{Q4} \\ H_a: & p_{Q1} \neq p_{Q4} \end{array}$

• •

--



Chi-Square	Value	DF	Significance
Pearson	292.26875	1	.00000
Correction	292.03442	1	.00000
Likelihood Ratio	295.10328	1	.00000
Mantel-Haenszel test for linear			
association	292.26731	1	.00000

The $\chi^{2}_{\alpha=0.05, \gamma=1}$ value is 3.814.

Since 292.269 > 3.814, reject H_o.

 $\begin{array}{ll} H_o: & p_{Q2} = p_{Q3} \\ H_a: & p_{Q2} \neq p_{Q3} \end{array}$

	Count	i quaf	TER	
RECORD		 2	3	Row Total
ACCESS	3	81339 	87016	168355 89.3
DEP LOSS	4	8753	11477	20230 10.7
	Column Total	90092 47.8	98493 52.2	188585

Chi-Square	Value	DF	Significance

Pearson	184.34333	1	.00000
Correction	184.14113	1	.00000
Likelihood Ratio	185.00642	1	.00000
Mantel-Haenszel test for linear			
association	184.34236	1	.00000

The $\chi^2_{\alpha=0.05, v=1}$ value is 3.814.

Since 184.343 > 3.814, reject H_o .

Chi-Squared Test

٩.

	Count		UARTER	Row
		I 21	4	Total
RECORD	3	81339	100054	181393 88.3
DEP LOSS	4	8753	15253	24006 11.7
	Column Total	90092 43.9	115307 56.1	205399 100.0

Chi-Square	Value	DF 	Significance
Pearson	604.56451	1	.00000
Continuity Correction	604.22425	1	.00000
Likelihood Ratio	613.23362	1	.00000
Mantel-Haenszel test for linear			
association	604.56157	1	.00000

The $\chi^2_{\alpha=0.05, v=1}$ value is 3.814.

Since 604.565 > 3.814, reject H_o.

 $\begin{array}{ll} H_o: & p_{Q3} = p_{Q4} \\ H_a: & p_{Q3} \neq p_{Q2} \end{array}$

ł



Chi-Square	Value	DF	Significance
Pearson	120.54121	1	.00000
Continuity			
Correction	120.39723	1	.00000
Likelihood Ratio	120.95478	1	.00000
Mantel-Haenszel test for linear			
association	120.54065	1	.00000

The $\chi^2_{\alpha=0.05, v=1}$ value is 3.814.

Since 120.541 > 3.814, reject H_o .

APPENDIX B DEP LOSS PROBABILITY ESTIMATES

.

4

		77	יר יייהגדנו	SCI LIER LOS	S PRCBAB	ILIT' ESTI	HATES USI	UL SIMPLE	PPCBABIL	IT' and I	SOTONIC R	EGRESSION		
								HCITHS IN	I THE CEP					
8	8C X	נר	I	2	ſ	7	S	ور	t.	8	e,	10	11	12
-		ANS	G.338	5.273	C.105	5.105	5:055	C.121	5.123	0.145	0.165	C02.0	0.249	0.215
~		Ч. Ч.	0.054	0.07	5.292	5.112	0.127	C.195	6.219	3.255	0.235	0.29	0.305	0.117
~		egys S	0.211	3.167	0.381	6:028	911.0	0.124	C.113	0.142	0.185	0.237	0.269	0.255
7	┢─	8	0.057	1.064	250.0	0.115	c.13?	C.2C3	6.223	3 .24 9	0.276	0.24	0.256	0.079
S	:	ž	ج. II	Ч Ч Ч	بر 1 2	ų ک ا	R II	N'A	بر <u>ب</u>	на	N. 11	¥/N	N:A	N.A
•0	-	Ĩ	C. C55	3.069	G. 373	0.000 C	680°0	191.5	E11.5	5.133	0.185	0.242	9.146	0.114
-		1 NG	C.C88	3.157	0.103	0.152	9.118	C.155	C.258	0.169	0.1	0.127	60.0	0.068
æ		8 H8 1	C.158	0.118	0.119	3.118	3.118	0.118	0.118	C.118	0.118	G.118	0.118	0.109
e ,		SFA	6.425	9.259	6.149	0.17B	6.196	0.248	C.216	0.238	0.279	16.0	0.386	0.365
10		GFÀ	C. 066	0.122	0.163	3.125	0.234	0.321	126.0	:159	0.419	0.354	0.43	0.158
=		SFB	0.667	г	1	1	-	7	7	1	1	I	1	1
7		3 FB	C. Cé4	0.095	££1.5	3.149	0.176	0.241	C:133	0.125	0.254	0.155	111.0	0.026
2	:	5F4	ы н К	4.11	ų n	R'11	Υn	ц й	¥ 11	нд	R'H	N'A	N∴N	H'A
2		354	C. C62	5.333	£££3	££€.¢	11111	1	1	1	1	1	1	0.062
2		LIFÀ.	0.1J	C1.3	5.5	3.5	3.571	0.571	0.571	172.5	J.571	0.571	-	0.5
2		1JE B	Ģ	13	5	o	()	U	c:	43	c	υ	0	0
5		SdH	3.055	9.08	86010	5.112	96010	5.113	C.136	C.172	c.133	0.13	0.086	9.102
18		FPS	C. CS3	3.088	0.103	5:25	3.137	C.142	C.193	1112	C.133	9.152	0.04	6.013
i î		члан Наге	5:05	C.C.C	C.132	3.12 5	_t.c	:.25	c.243	21212	161.5	C.233	0.182	0.102
20		BHB	5.CT3	5.12	2.1.2	C.217	5.233	5.233	:.233	2.23	5.233	0.233	0.265	0.265
21	•	HFÀ	0.11	5.25	c.25	3.286	c.285	:.333	::33	c.333	1	1	1	5.5
32	•	HFB	c)		، ،	1	1	1	1	1	1	1	-	ı
		indica	tes 50% p	orobabilit	estimat	Les calcul	ated with	data fro	mall fou	r quarter	s, using	isotonic	regrassio	-
	:	indica	tes BCNes	s that are	inot raci	ruited								

			QUARTER 1	EEP LOSS	PROBABIL	IT' ESTIM	ATES USIN	SIHPLE I	PRCBABILIT	" and IS	DTCNIC REC	CRESS 1011		
								HUITHS III	THE CEP					
Ê	eo Xo	51	7	2	£		S	۰.	r.	8	¢.	10	11	12
-		SHR	3.216	2.355	5.313	C.104	5.2	0.183	3.169	0.146	9.138	0.14	0.284	0.38
~		545	3.052	6.372	5.103	C.122	0.179	0.206	≎.22é	C.19	0.18	0.238	2.247	0.144
~		aits	5.114	0.152	5.17	1.0	5.148	5.154	3.132	1111	C.152	0.165	0.247	0.471
-7		SMB	0.054	0.053	3.08	3.102	C.126	5.196	::113	2.176	C.302	0.151	0.128	0.124
S	:	SH4	5.2	4.11	ب ۲	¥.11	ς Ξ	Қ .1	¥.11	Ч Ч Ч	A' H	N.'A	N/A	¥∵N
•		544	0.048	12.0	0.0é	0.617	0.079	0.167	0.208	241.3	0.122	9.05 <u>9</u>	0.059	0.059
۲.		(BIC)	2.027	0.123	1-11-6	C.118	0.102	C. C67	1.0.G	6.013	0	0.156	0.098	0.081
6	ŀ	17MB	9.1.0	911.0	811.C	C.118	5.118	0.118	3.118	3.118	0.118	0.118	0.118	0.108
o.		SFA	0.344	116.0	7	6.422	2.333	0.3	0.227	0.278	0.244	0.237	0.395	0.537
2		GFA	0.06 1	0.118	5.1.5	6.226	6.204	0.347	9.277	0.272	0.405	0.298	0.29	0.148
=	ŀ	SFB	3.657	-	-	-	-	-	-	1	1	ı	1	1
12		GFB	0.05	0.136	0.179	641.0	0.201	C.224	5.213	102.0	0.16	0.167	0.057	0
2	:	1-1S	4 2	ю С	ų Ξ	بر :: ت	ы à	к. П	ж П	ЧÄ	A.U.	N:A	H:A	ы А
=	ŀ	3F4	C. C62	0.333	5.333	C.333	:.333	-	1	1	ı	1	1	0.062
15		IIFÀ	c.13	5.1×	3.5	c. S	0.571	0.571	:.571	0.571	0.571	0.571	1	5.5
۲	ŀ	11FB	0	: 1	IJ		ι,	IJ	ι,	()	υ	o	c	0
1		5dH	0.057	: • : · :	5. CB	::: :	ι ι ι, ι,	C. C 18	:.14é	:.141	5.125	C. C53	C.107	2.152
18		FFS	C. 045		5.53	6.026	0.052	0.07S	C.C83	:.12?	0.135	0.135	C.133	0.133
15		¥11H	0.065	0,056	60110	:151.5	5.123	3.148	5.15 ⁷ -	:.1	C.1	t-60-0	c.12	111.5
20	ŀ	ныя	0.679	0.12	5.12	512°S	5.233	:.233	2.233	c.233	c.233	G.233	G.265	3.2éS
21	·	HFÀ	C.11	0.25	2.25	C.285	5.28é	3.333	:.333	:.333	1	1	-	5.5
22	ŀ	HFB	o	ι,	0	1	1	-	-	-	1	-	~	-
	<u> </u>	indice	ates 80%	probabí l il	t' estimat	tes calcul	ated with	data fro	wall fou	r quarter	s, using	isotonic	regressio	c
	:	indice	ates BOXe:	s that are	roer roei	ruited								

			QUARTER 2	LEP LISS	PROBABIL	IT' ESTIN	ATES USIN	SIMPLE 1	PROBABILI	l' and IS:	OTCHIC RE	CRESSION		
F	F	$\left[\right]$						II SHIICH	I THE DEP					
	BO X	3	1	2	ŕ	7	S	م،	2	9	01	10	11	12
-	F	SHA	G.148	0.34	0.116	101.0	111.5	11.0	c.11	C.116	0.244	0.311	9.436	0.355
~	┢╴	SHE	9.055	;.Csé	3.136	541°C	811C	C.268	C.182	5.23 7	3.278	0.302	0.368	0.136
~		SHB	c.125	C.182	9.104	0.103	5.136	c.117	0.096	0.103	C. 283	0.39	0.476	0.471
-		BND	0.657	C. Céé	0.126	0.134	C. 151	C.233	0.229	2.197	0.049	0.2	0.12	0.077
~	† :	5H4	14. N	4-11	ų.11 الم	¥.11	R H	K 11	ڊ <u>ب</u>	11 Å	¥. N	A.N	N N	N°A
••	1-	ž	9.052	0.076	120.0	9.108	3.029	C. C64	C.1C9	C.113	C.063	0.116	0.116	0.116
-	┞	NBI	0.087	C.2C3	3.169	0.065	3.057	C.077	0.116	C.2	0.174	0.14	160.0	o
•	 .	aren	0.108	0.118	911.0	611.C	0.119	5.118	0.118	C.118	0.118	0.118	0.118	0.108
а.	┢╴	SFA	201.0	221.0	C.105	0.139	3.18?	C.214	C.185	C.265	G. 302	0.449	0.6	0.475
10	\uparrow	GFA	0.071	6.117	3.189	t61.5	0.273	C.292	5.28	C.416	9.4	0.372	0.218	9.14
=	ŀ	SFB	0.667		-		1	1	1	1	1	1	1	1
2	╞	GFB	9.068	C. C2S	0.136	5.264	5.213	5.223	0.312	0.306	c.25	0.106	0.019	0
=	:	5F4	R.R	¥. 1	بر :: ا	к II	Ϋ́	Ч II	स् 11	цλ	4. N	¥. N	N A	۲ . N
=	 -	GF4	3.062	0.333	0.333	::133	£££.5	~1	1	1	1	1	1	G. Cé2
5	† -	LIFÀ	61.0	C1.0	5.5	0.5	2.571	172.5	C.S71	C. 571	a. 571	0.571	1	û.5
16	† -	NFB	G	0	o	6.7	O	IJ	c)	IJ	c	0	Ü	J
1		SdH	0.053	0.0.0 0.0.0	2 11 7	0.1	2.282	601.0	911.5	C.163	0.143	0.152	10.0	G.054
81	\uparrow	FPS	0.072	5.123	1- 1- 1- 1- 1- 1- 1- 1- 1- 1- 1- 1- 1- 1	0.082	111.0	141.0	C.163	c.157	:.151	5.077	0.077	c. c77
ćI	1-	ЧH?	0.051	C. 563	5.125	5.122	11.2	C.183	3113	:.125	C. 121	0.167	0.105	0.031
2	ŀ	BNH	61010	:.12	2.12	:1212	5.233	:.233	5.233	C.233	:.233	0.233	D.265	C.265
17	ŀ	HEY	11.0	2.25	0.25	5.295	5.285	133	1113	5.33	1	1	1	c.5
R	. .	HFB	0	••				1	ι	1	1	1	1	1
Γ														
	1.	indica	tes BCX	propabili	emijse (j	tes calcu	lated with	i data fro	m all fou	r quarter	Guisu ,s'	i sotoni c	regressic	ç
ſ	:	indica	stes BCXe	s that ar	P not raci	ruttaci								

		410	RTEP. 3 CE	RY SSCI 4	BABILITY	ESTIMATE.	s criisti s	ICAN BURN	BABILIT' a	ICLCSI Put	UIC RESRE	NOISS		
								HOLIT	HS III THE	CEP				
	BO. NOE	CL	1	2	5	7	Ś	Ŷ	7	8	٥,	10	11	12
		SHÀ	C.328	0.17	0.078	tot o	5.087	C.126	9:1:8	3.219	C.264	0.366	166.0	0.202
~		410 1	5.05B	5.279	1-C - C	:114	0.113	0.205	5.216	:.252	0.343	0.296	96.0	101.0
~		SHB	S.2	101.0	0.372	16512	0.097	C.122	0.25	5.162	C.289	776.0	0.5	0.233
-		348	3.302	3.265	5.282	5.138	:155	C.189	0.205	0.281	C. 364	0.282	161.0	0.069
~	:	t HS	1. Z	¥.11	ц н Н	R.11	۲۱ Å	۲ H	H.A.	ЧÄ	A 11	Y, N	NZA	A. N
۰		Ř	0.06	140.0	180.0	0.117	C.119	561.3	0.167	5.18	0.205	0.27	0.27	0.27
6		र्मन्न	0.084	0.142	3.227	101.5	C.1C7	31.5	C.306	3.25	0.25	0.173	0.132	0.111
0	ŀ	8481	5.138	911.C	3.118	2.118	C.118	9119	G.118	3.118	0.118	0.138	0.118	0.108
с.		SFÀ	0.375	0.18?	c.143	0.178	C.197	516.0	9.375	0.378	0.4	0.5	0.533	9.319
5		GFÀ	6.074	0.151	0.151	0.189	2.2	5tE-3	0°36	5.426	C. 385	6.0	0.5	0.153
=	·	SFB	0.667	-	1	1	1	1	1	1	1	1	1	1
12		GFB	0.067	0.284	5.123	5.129	C.124	C.218	0.138	5.224	0.258	0.283	0.082	580.0
13	:	SF4	¥:R	Ψ. 12	н А	нΑ	ЧЧ	ч н Ч	ц н И	ц н 1	R.H	¥. N	N∛A	н.А
2	ŀ	3F4	3.652	6.333	5.333	5.333	666.5	1	1	-	1	ı	1	J. uč2
15	·	LIFÀ	51.5	0.13	5.5	c.5	1.571	1-2.5	1:5:0	:.STI	0.571	0.571	1	o.s
· <u>°</u>	•	Nr.B	o		. ,	ι,	0	ί,	() 	0	C	υ	0	σ
1.		SAH	0°02	\$01.5	90 1 0	301.5	:.125	5.13	2.152	C.2	c.20S	0.124	0.145	:111
18		SdJ	C. C52	30.0	:11:	3.127	:.21 ⁻	161.5	:.235	:.22	0.21ó	0.073	c.079	5.0.C
13		-SHK	0.052	111.5	:153	÷11;	2.162	£273	5.223	C.26	0.347	C.281	c.25é	3.156
23	·	внн	51515	2112	2.12	5.217	:.233	::233	5.233	2.233	5.233	0.233	C.265	3.265
17	•	HFA	C.11	3.2 5	0.25	2.285	C. 296	5.333	c.133	2.333	1	-	1	3.5
22	·	HFB		¢1	••	1	-	1	1	l	1	1	1	1
		indica	1 XCH sel	robabili.	; estimat	es calcul	ated with	data fro	m all fou	r quarter	s, using	isotonic	regressio	c
	:	indica	ites BCXes	s that are	e not rest	uited								

			QUARTER 4	I CEP LISS	PECBABIL	IT' ESTIN	ATES USID	3 SINPLE	PRUBABILI'	T' and IS	CTCHIC RE	CHESSION		
								HUTHS II	I THE CEP					
	80.1.08	C C	I	~	۳.	7	5	ە.	(5	8	ø,	10	11	12
-		સમક	6.276	5.752	5.203	C. 245	2.165	391.5	0.241	275.5	0.262	162.0	0.237	0.193
~		KH5	3.35	C.265	660.0	C.103	121.3	5.18	5.264	3.328	0.376	0.313	0.287	0.107
~		BHS	C. 333	C.613	5,532	3.253	0.23	:.255	2.282	3.4Ró	0.352	0.261	0.251	9.225
-		9HB	0.055	4 L 3 - 2 2 - 5	0.003	hotro	C.125	197	3.288	6.313	16.0	0.301	é£2.0	0.068
\$:	5H1	بر 1	بر 2	۲ ت	Ч I	بر 2	Ϋ́	Ч I	२ ॥	N:A	¥∵N	K .:N	∀ ∴N
0		145	0.07	C.071	80°0	\$11°	5.154	111.2	2.152	:.216	0.117	601.0	0.071	0.026
6		(MG)	0.07	G.162	3.096	0.20 2	C.122	2.167	5 - 5 V	- 34	0.2	té0.0	10.0	0.06
<i>.</i> ,		1 MB	C.1C8	0.118	911.0	3.118	C.118	5.118	C. • 19	0.118	0.113	0.118	0.118	0.108
e.		SFÀ	0.679	0.75	Q.634	0.2	0.39	2.238	61t°C	3.36	tt£.0	66Z'O	0.35	0.327
12		3FÀ	c. 255	C.11	5.152	3.188	:.252	21307	c.328	5.353	0.463	9-396	0.446	171.C
=	·	SFB	0.667	-	1	-	1	1		1	t	1	τ	ı
2		ĠFB	C.072	5:2.0	£113	c.15	C.225	3.28T	C.136	C.136	0.127	0.132	0.063	760.0
2	:	SF.	بر 2	ج = =	к 1	× 1	ų ۲	Ϋ́	Ч Ч Ч	нλ	¥. N	A:H	N .A	¥. 11
2	•	GF4	3.262	0.333	5.333	5113	0.033	1	-	1	1	1	ī	0.062
2	·	11FÀ	£1.5	c.13	5.5		:.5-1	172.5	C. 571	1:5:0	172.0	0.571	1	3.6
10.		1159	(')	()	()	0	ن ا ا	0	۰,	o	G	υ	0	υ
1		5 dH	5275	C.C.T	3950.0	:139	C 563	C.13 [°]	5.053	C. CT 8	9.08	0.386	0.08	5.07é
		544	21010	0.058	5.118	180.0	c. c5	C.C97	tic:t	5:15	0.068	υ	0	ي ا
÷		રમ:	101.0	3.152	5:135	c.125	t_1.5	:.352	2.325	: 299	C.323	9.2 6 8	0.142	C.103
		Нив	C.C72	C.12	2.12	:1Z12	5.233	: [2]	5.233	5,223	:.233	6.2.3	0.265	C.265
51		Η£λ	11.1	2.25	3.25	3.2Ró	2.285	5,333	5.333	:: : :	1	1	ĩ	c. 5
22	•	HF ð	,	4,5	43	7	7	-	1	1	1	1	1	1
		ındicə	ites BCX p	robabilit	temi tse	es calcul	ated with	data fro	Mall fou	r quarter	Euisn 's.	isotonic	regressic	۲.
	:	indica	sexual setu	s that are	not recr	uited.								

APPENDIX C 95% CONFIDENCE INTERVAL UPPER VALUE PROBABILITY ESTIMATES

•

• • • •

				ALL TIAR	TEPS PRCB.	ABILIT'S	F CEP LOS	SS UPPER	A VAULE DI	F 258 C.I				
								HI SHTICH	THE CEP					
1.08				~	-	-	S	ه.		8	o.	10		12
-		SH:	0.37	c. 296	5.12	3.112	5.107	0.13.	561.0	0.154	C.174	0.213	0.258	0.225
~		я <mark>е</mark>	0.356	0, 073	1:1:1	5.117	CC1.3	532.C	C.236	3.277	C.322	0.32	0.338	0.129
~		8 73	0.244	161.5	2.236	111.5	161.0	9110	0.123	5.154	C.2	0.255	0.286	C.273
-		80	0.06	5.262	5, 2, 3	c.124	0.152	0.231	2.258	0.295	C.335	567°C	0.323	0.095
s	:	Īŝ	۲ 12	۲ 1	بر 2	ų z	ب ت	κ =	R 12	۲ ۲ ۲	R N	R. N	N:A	N.A
••		Ř	0.05	51212	3.388	0.13	C.12 ⁻	0.272	2.199	3.255	C.324	0.395	0.272	0.189
(-		Line,	9. J29	C.185	128	0.198	0.159	0.239	C. 367	2.26	0.166	0.209	0.153	0.101
e		8	0.165	C. 277	1-1-0	0.177	0.177	571.C	3.177	5.177 C	0.177	0.177	0.177	0.165
<i>c</i> .		SFA	0.502	2.312	0.181	C.202	0.221	3.279	3.245	3.332	0.314	0.35	0.429	0.416
21		SFA	0.071	161.5	5.174	0.256	5.242	6+C-0	0.357	0.403	0.472	0.413	0.501	0.183
::	ŀ	SFB	0.753	-	-	-	-	1	1	1	1	1	T	1
2		8.3;	0.07	2.105	3.148	C. 1 66	C.122	1:2.0	0.257	5.268	0.358	0.248	0.189	0.043
ñ	:	SF1	л 2	بر =	ų Ξ	بر =	Ϋ́Ξ	ч Ч	ب ۲	ب <u>۲</u>	۲۱. A	Ы. Ы	К. N	¥. N
2		5F4	3.106	514°C	é 1 f 13	:11:	c. 11 >	1	1	1	1	1	1	0.136
15		iFA	161.0	191.5	0.521	145.5	2.001	0.661	5.661	J. 661	0.661	0.661	1	0.591
16	ŀ	1158	.,	. ,	11	6	••	••	•	0	C	c	0	Ċ,
-		SAH	3.258	ر بر به ر ۲۰۱۲ - ۲۰	c113	C.133	5.118	t1.0	¢.176	2.237	C.21	3.228	0.178	0.14
18		FFS	3.062	11.5	5.14	c.13	2.187	1:2:5	2.2 ⁶	2.266	0.255	9.28	0.11	0.037
2		가분	2.264	с. Ст 	2.149	c.153	704-0	::::	2.122	5.345	0.216	0.413	5.284	c.145
~		874H	0.128	5.179	5.1.S	C.232	:.31	C.31	с. M	16.5	c.31	3.31	C.346	6.346
51	ŀ	HEÀ	2.167	2,329	2.329	c 31.2	2.360	dtt:	étt:2	:115	1	1	1	1:5.5
2		₽J#	IJ	0	۰,	1	1	-	1	-	-	-	-	-
		indicate	s BCN Fre	chabilit;	estimates	s calcula	ted with	data from	n all fou.	r quarter	s, using	isotonic	regressic	ç
	:	indicate	s BONAG t	chat are i	not recru	itəd								

				1972	TER 1 PRC	BABILITY	CF EEP LO	SS UPPE	R יאנייב ז	F 250 C.I				
								H SHTHS H	THE CEP					
2			-	2	'n	7	S	,o	t:	8	٥,	10	11	12
-		15	0.322	3.463	5.323	61.5	5.288	:.232	191.5	2.156	0.147	0.153	0.315	ē.419
~		ų K	730.C	612.5	2.112	0.133	2.207	:.244	3.27	162.2	9.222	3:2 9€	0.30g	0.173
~	Γ	8 9 65	3.228	0.274	5.278	C 61 ° C	c.23 ⁻	9-1-B	C.153	č.157	0.168	C.193	9.304	0.5é
-		86	2.36	1.00.0	5:000	0.134	C.174	2.272	0.215	5.258	0.44	0.247	0.233	0.169
~	:	Ĩ	Ч Ч	بر 11	Ϋ́	بر ا	.ч 2	R. 11	К 12 Г	ЧЧ	ΥN	¥. N	K.N	N.'A
ه.		Ř	c.057	a.ce3	£90.0	0.051	3.165	¢.)	0.353	0.274	C.239	0.143	0.143	0.143
•		1	2.122	0.185	212.2	0.233	C.21	0.156	0.11	0.054	٥	0.282	0.204	0.148
ø	ŀ	8	0.165	0.177	1110	24110	5.177	3.177	0.175	2.177	0.177	0.177	0.177	0.165
c.		SFÀ	0.514	¢.514	5.61P	J. éĉé	2.502	0.345	0.274	0.313	G.284	é0E.0	0.502	0.689
27		117	0.075	2.136	\$61.5	5.257	2.248	0.417	2.365	6.354	C.S17	0.429	0.45	0.206
=	ŀ	SFB	C. 753	-	-	-	~	7	-	-	1	1	1	1
2		e .,	с. с е́	C.128	C.213	C.182	C.268	116.2	j. 35	C.348	162.0	٥.3	0.14	0
12	:	SF 4	4 Z	بر 2	ب 2	بر 2	К.П	بر =	بر =	¥ 11	A'N	२ . त	N:'A	4. II
	•	727	3.126	611.5	5.419	511.2	5.412	1	-	1	1	٦	1	0.106
2	ŀ	にす。	151.5	161.2	1:5:2	142.2	3.661	5. óč 1	3.661	1.661	0.661	0.661	1	0.591
2	ŀ	Ę	Ċ	، ,	.,	r.)	o	.,	13	() ⁻	o	0	0	0
12	Γ	HE C	190.0	a. 	5:1:3	141.5	611.5	5.033	2.246	2.2-2	C.243	0.133	C.218	c.238
2		FFS	3.262	2.122	:.153	0.011	161.5	0.162	5.182	:12.3	5.257	0.257	0.255	0.255
2		is ₩	د.: <u>-</u>	7.40.0 0	:116	5.219	2,247	3.275	5.3	5.2 5 ¹¹	5.207	8át.0	0.236	5.233
14	ŀ	874	5.12B	€_1.5	5.17	2:292	16.5	16.5	16.5	11.2	0.31	0.31	C.346	C.346
7	ŀ	HFÀ	2.167	5.325	2.322	0.362	2-362	5.412	c11-2	:115	1	1	1	192.5
2	ŀ	нге	.,			-	-	1	1	1	1	1	1	-
].	indica	ates BCN	probab: 11	ty estimat	tes calcul	ated with	i data fro	m all fou	r quarter	s, using	isctonic	ragressio	n
	:	Indice	stes BCKa	s that are	Pot raci	ruited								

				4.5	PTER 2 FK	CBABILITY	JE CEP L	SSS CHEE	R TALVE C	F 254 C.I				
								HOUTHS II	I THE CEP					
2			-	~	^	-	s	۰۵	ı	8	٥,	10	11	12
~		5965	0.243	3.467	1.161	5.132	c.124	2,122	5.12	161.5	C.288	0.35 <u>9</u>	0.491	1.0
~		50	9,058	5.572	C.12	6.17	0.203	0.238	:.21	3.28	C.346	0.376	0.46	0.163
~		8065	5.24	CIC .:	c.15	3.127	3.157	161.2	3.138	:129	0.372	964.0	0.563	0.56
-		8	C. D62	C - 0 - 1	2.155	c.173	c:1:5	3:295	162.0	3.236	C.115	666.0	0.236	0.119
5	:	775	¥. N	بر <u>ا</u>	Ϋ́		11 4	ч г	ب =	л 2	K. 11	N A	N.'A	N'A
•		Ř	0.059	C.336	0.09	0.175	3.668	c.152	:.221	0.226	0.15	0.231	162.0	0.231
~		1 MA	6.136	6.228	C. 3C3	0.153	0.14	2:172	5.231	eter::	C.31	0.264	660.0	0
•	ŀ	1	ĉ. 16S	C 1_2	- - 	6-1 -	5.177	0.177	0.177	:.177	0.177	0.177	0.177	0.165
-		SFA	C. 333	116.5	2.185	с. 1 è	0.223	C. 25	5.223	3.345	C. 44	0.588	0.762	0.63
2		121	0.081	1111	5.222	236	516.5	e. 355	5113 C	C.526	0.562	0.517	0.366	0.197
=	ŀ	SFB	0.753	-	-		-	٦	-	1	T	1	l	1
2		845	610.0	c.11.5	91°C	5.34e	:.2?é	2.3óR	817°C	17 1 .5	0.405	0.216	0.068	0
2	:	74.	۲ 2	й н Х	<i>ң</i> 2	ڊ <u>ت</u>	۰۲ ۲	بر 2	۲ II ک	Ч Ч Ч	R N	R. 11	N:A	R. 11
=	ŀ	T.A.	C.106	411-5	5.412	c.112	:11:	1	1	1	ſ	1	I I	0.106
15	·	185A	191.2	191.5	16312	C. 5?1	101:	2. é é l	100.2	3.651	0.661	0.661	t	165.0
19	ŀ	1158	Ċ.	11	c,			0	()	Ģ	C	0	0	a
		Sah	5.059	5 e 2 · 2	118	C+13	561.2	: 1 - :	:12.5	5.273	C.268	9.2.8	1:1:0	0.127
6:		544	160°C	0.175	0.1é1	5.18	5.223	3.260		C.28 ⁻	0.279	9.172	0.172	5. 1 72
2		H::h	c. c5#	C. C. B. S	2.1.2	1.222	1 _ 2 * 2	125.5	5.233	512.2	C.238	c.3	312.C	2:232
	·	637 8 4	2.129	5.1.5	6_1°2	2:252	:.31	10.5	16.5	C. 31	::31	0.31	3.346	C.346
5	·	HEÀ	C. 16 ⁻	5.329	525.5	:96.J	::::	ett.2	ett 12	ilt':	1	1	1	0.521
7	·	HFB	()	"		ı	1	1	1	1	1	1	1	1
ĺ		indica	ites BIX p	rchabilit	eet imat	es calcul	at statut	i Jata fro	mall fou	r quarter	rs. using	isotonic	regressio	ç
	:	1001Ca	Seal & Seal	that are	not rec	uited.								

									101 101				
							UT SHTTIC	THE CEP					
		-1	2	. .	7	s	é	7	8	o,	10	11	12
-	:5 1 :5	2.365	241.2	262.5	5.116 5.116	5.027	:.15	5.24	5.2SF	0.305	0.442	0.301	C.217
~	ŝ	C. CúJ	÷8010	121.2	0.123	0.122	262.2	:.253	3.295	5.391	9.363	C. 464	0.123
~	SHE	2.235	2.125	2.28é	2.106	611.5	C.171	2.32é	5.218	2.361	0.508	C.é17	0.25 4
-	980	:.:	\$1010	2.101	5.157	2.176	0.239	2.275	1:1::	0. 48	0.423	0.252	0.101
:	ī	K. 11	بر =	ڊ 11 ا	4.H	بر 2	ų.1	к 11	ې ۲	R 11	4. N	K. N	N A
	ž	C.071	c. c55	5.102	3115	0.12"	0.252	C.3	0.317	6 1 6.0	0.429	C.429	D.429
	1967	9.101	241.2	:.351	0.152	5.173	0.278	C.471	0.405	0.405	0.3C8	0.253	0.203
•	8941	C.165	2.1 .2	5.177	24172	5-1-5	5.177	111.5	5.177	0.177	0.177	0.177	0.165
с.	SFA	5.472	3.241	0.179	91010	G.233	3.389	C.487	0.478	0.515	0.679	0.712	0.391
10	123	C. C86	5.172	C.17	0.2CB	0.228	5.40é	0.471	2.482	0.478	0.427	0.668	0.199
=	SFB	6.753	-	-	-	-	-	-	1	1	1	1	1
2	ß	0.079	2:1:2	2.145	0.155	2,155	5.293	5.227	146.5	C.415	0.444	0.18	0.083
:	SF4	K 1	× =	ų Ξ	ų I	بر =	ц А Ц	¥.11	۲ A	A:U	N.11	Υ, N	11 - À
•	ī,	3.106	÷1112	514.5	51 1 13	51413	7		1	1	1	1	0.106
•	I IF À	191.5	141.2	0.521	165.2	5.601	5.66 1	1.661	3.661	0.661	0.661	1	C.5?1
•	821 1	IJ			D.		(1	o	Ø	Ċ	0	Ð	o
	SAN	5.CS7	2.12÷	2.132	111	2.165	111.5	122.2	5 11 5	5 1 610	0.336	0.271	C.195
18	544	C.C73	2.12÷	₽ć1°3	0.204	51322	5.253	j.j⊺ð	2.366	C.36J	0.176	0.176	c.176
9. 	S/84	1.261	11.5	615.2	161.2	5.222	5.3C÷	C.368	:11:5	0.517	c. 442	0.412	2.282
•	87 H	C.128	C.179	5.1.5	2:232	16.5	16.5	16.5	16.5	16.5	C.31	0.346	C. 346
17	HFÀ	2.167	0.322	5.323	:.35?	5.362	C 11 5	51t*S	51 1 5	1	1	1	1 é S · C
?	нгв	ι.	4.5	13	-	Ţ	1	ſ	1	1	1	1	1
. 	Indica	tes BCX p.	obabilit.	stimat	es calcula	ted with	data [rom	all four	quarters	i, using i	sotonic r	egression	
:	Indica	tes BONes	that are	not recru	ited								

	ļ			94:12	TER 4 PR	BABILIT'	CE CEF LO	addit: SS	CR TAULE 2	F 258 C.1				
Γ	Γ							HOLTHS II	I THE CEP					
	Γ		-1	~	ĥ	-	S	••	t' ·	8	ć	10	11	12
-		in the second seco	0.552	C.823	3.228	3.332	c.222	C.243	C.32	51t°C	10.294	0.245	0.245	0.204
~	Γ	援	C. 055	C.C72	2.107	5.11	161.2	3.2	C.299	c.374	0.434	0.363	0.336	0.124
~	Γ	e Ko	0.432	5.763	117.5	0.428	\$66.3	3.375	c.423	C. 648	0.414	0.286	0.269	0.251
7		8	0.061	a) `) } }	2.104	3.115	5.144	C.245	5.365	101.0	0.417	0.407	0.339	0.08 ⁹
~	:	i i i	K.11	Ϋ́.	ų.u	بر =	¥. 12	х =	يد. 1	к 11	Y. N	A.N	K.N	N A
•0		ě	C. 084	č. C85	911.0	2.197	2.267	2.268	5.2B	C.363	0.232	0.221	0.163	0.075
1.	Γ	ž	0.136	0.221	121.5	0.262	2.18J	0.3	c.452	12.0	545.C	C. 198	0.189	111.0
•	ŀ	8	C.165	C.177	6.177	2.177	5.177	5.177	5.177	3.177	0.177	0.177	0.177	0.165
e .		SF à	C.846	6.C	617.5	646.5	3.565	3.39	C. 596	3.532	0.443	6.353	t-0	0.418
1:	Γ	150	6.064	c.124	0.177	3.236	2.276	3.35	C.387	C.433	0.552	68t O	0.542	0.214
=	ŀ	es S	0.753	-	-	-	-	-	1	1	l	1	*1	1
2	Γ	83	5.087	¢.112	0.137	C.185	5.2 7	C.382	:.238	C.259	J.24 6	0.253	0.146	0.066
=	:	SF4	K-11	بر =	ų I	۲ II	ų I	Ч. П.	بر =	ыа	¥:1	N.A	N.'A	Y. N
2	ŀ	ī,	9.106	5119	é11°2	511.5	5113	1	-	1	l	1	1	0.106
ŝ	ŀ	IRA	1:13	161.5	145.3	145.3	3.661	C. 66]	2.661	3.661	J. 661	0.661	1	1:5.0
:1	ŀ	1478	υ	U.	u	()	ι,	.,	••	()	0	D	٥	C
<u>'</u> _		SaN	C. Jéé	160.0	C.126	1.183	C.155	1:2:5	5.124	C.174	111 C	0.186	0.177	0.134
e,		S43	5,059	56010	161.2	2.148	5.13	5.203	C.138	2:132	0.158	6	0	••
0. I	Γ	ц. Mari	5.123	2:122	:153	2.165	0.22?	5.t-3	C.466	C.463	61.0	n. 426	0.267	9.17
::	ŀ	8H8H	€.12 0	5112	2.172	2:232	16.3	0.31	16.5	c.31	16.0	0.31	0.346	3.346
2	ŀ	HFŻ	0.167	:.329	2.322	:.369	5.362	6 11 3	étt's	:415	1	1	1	1:5:1
2	ŀ	HFB	e	.,	1)	1	1	1	1	1	1	1	1	1
		indica	Las BCN L	orchebilit	:; estimat	tes calcul	lated with	idata (ro	w all fou	r quarter	s. using	isotonic	regressio	ç
Γ	:	indica	tec BUXe	s that are	not rect	ruited								

APPINDIX D FORECASTED NETDEP MATRIX

7.20

.

· · ·

		AC	CESSION	MONTH		
<u>BOX</u>	OCTFY1	NOVFY1	DECFY1	JANFY1	FEBFY1	MARFY1
SMA						
GMA	3955	2492				
SMB						
GMB	1703	1952	458	1164	430	
GM4	121	126				
NMA						
NMB						
SFA						
GFA	698	799	133			
SFB						
GFB	300	344	80	189		
GF4	21	7				
NFA						
NFB						
PS	511	317				
HMA	304	348	81	18		
HMB						
HFA	53	45				
HFB						
	APRFY1	MAYFY1	JUNFY1	JULFY1		
SMA			3279	133		
GMA						
SMB						
GMB						
GM4						
NMA						
NMB						
SFA			560			
GFA						
SFB						
GFB						
GF4						
NFA						
NFB						
PS						
HMA						
HMB						
HFA						
HFB						

APPENDIX E

A. MONTHLY CONTRACT REPORT

.

		<u>CONTRA</u>	CT MONTH		
BOX	OCTFY1	NOVFY1	DECFY1	JANFY1	FEBFY1
SMA		4063		4020	
GMA	1146	1951	6384	961	3768
GMB				1	
GM4		1186			
NMA	2648	156			2616
SFA	250	372		575	
GFA	1452			1343	
GFB	889			822	
MONTHLY					_
CONTRACTS	6384	7728	6384	7728	6384
EXPECTED					
ACCESSIONS	5871	6812	5958	6737	6028
RECRUITER					
CAPACITY	6384	7728	6384	7728	6384
SHORTFALL					

		CONTRA	CT MONTH		
BOX	MARFY1	APRFY1	MAYFY1	JUNFY1	JULFY1
SMA			649		
GMA	4666	62	1453		3218
GMB	666	932	5154	6720	130
GM4					1597
NMA	1388				
SFA		251			
GFA		586			760
GFB		359			465
PS		4531	808		214
MONTHLY					
CONTRACTS	6720	6720	8064	6720	6384
EXPECTED					
ACCESSIONS	6338	6183	7457	6310	5826
RECRUITER					
CAPACITY SHORTFALL	6720	6720	8064	6720	6384

77

772

	CONTRA	CT MONTH	BOX
BOX	AUGFY1	SEPFY1	TOTAL
SMA	4839		13577
GMA	3561	4510	31680
GMB		1548	15151
GM4			2783
NMA			6808
SFA		326	1774
GFA			4140
GFB			2535
PS			5553
MONTHLY			
CONTRACTS	8400	6384	
EXPECTED			
ACCESSIONS	7378	5885	
RECRUITER			
CAPACITY	8400	6384	
SHORTFALL			

-

B. BASIC TRAINING SEAT JTILIZATION REPORT

ACC	CONT	ACC	NETDEP	TOT	REQ ACC	# UNF	😵 FILL
MONTH							
OCTFY1			7666	7666	7666		100
NOVFY1	2598	2424	6430	8854	8784		101
DECFY1	1386	1307	752	2059	2059		100
JANFY1	4117	3870	1371	5241	5241		100
FEBFY1	5958	5529	430	5959	5959		100
MARFY1	4954	4679		4679	4679		100
APRFY1	4666	4400		4400	4400		100
MAYFY1	6203	5808		5808	5808		100
JUNFY1	2757	2515	3839	6354	6354		100
JULFY1	8051	7523	133	7656	7656		100
AUGFY1	8676	8015		8015	8015		100
SEPFY1	10009	8397		8397	8397		100

LEGEND:

ACC MONTH = month contract is programmed to access CONT = number of new contracts written to mature in a given month ACC = number of new contracts that are expected to assess in a given month NETDEP = number of previously written contracts expected to assess in a given month TOT = total number of new and previously written contracts expected to assess in a given month REQ ACC = required number of accessions in a given month # UNF = difference between REQ ACC and TOT

% FILL = percent of required accessions expected to achieve

C. FY94 MILITARY SERVICE MIX REPORT

	CONT	ACC	NETPEP	TOTAL	DA MISN
PS	4531	4245	828	5073	5004
NPS	54844	50221	19793	70014	70014

LEGEND:

PS = prior service

NPS = non-prior service

CONT = number of new service category contracts written to mature in FY94

ACC = number of new service category contracts that are expected to assess in FY94

NETDEP = number of previously written service category contracts expected to assess in FY94

TOTAL = total number of new and previously written service category contracts expected to assess in FY94

DA MISN = total number of service category accessions required to assess in FY94

D. QUALITY AND GENDER MIX REPORT

, ۱

ACC	NETDEP	TOTAL	& ACHVD	<pre>% REQUIRED</pre>	& SHORT
				-	
47570	18944	66514	95.00	95.00	
34012	12898	46910	67.00	67.00	
1126	275	1400	2.00	2.00	
7133	3229	10362	14.80	14.80	
	ACC 47570 34012 1126 7133	ACC NETDEP 47570 18944 34012 12898 1126 275 7133 3229	ACCNETDEPTOTAL475701894466514340121289846910112627514007133322910362	ACCNETDEPTOTAL% ACHVD47570189446651495.0034012128984691067.00112627514002.00713332291036214.80	ACCNETDEPTOTAL% ACHVD% REQUIRED47570189446651495.0095.0034012128984691067.0067.00112627514002.002.00713332291036214.8014.80

ACC = number of new category contracts that are expected to assess in FY94

NETDEP = number of previously written category contracts expected to assess in FY94

TOTAL = total number of new and previously written category contracts expected to assess in FY94

% ACHVD = percent of required category accessions expected to achieve

% REQUIRED = percent of category accessions required to achieve

% SHORT = difference between % REQUIRED and % ACHVD

E. FY94 TO FY95 REPORT

	FY2 ACC	FY1 TO FY2	% ACHVD	% REQ	% SHORT
CATEGORY					
PS	3001	893	29.75	29.75	
NPS-A	48252	18456	38.25	38.25	
NPS	72017	21425	29.75	29.75	

.

FY2 ACC = number of FY95 category accessions FY1 TO FY2 = number of FY94 category contracts expected to access in FY95 % ACHVD = percent of required category FY95 accessions that . are contracted in FY94 % REQ = percent of category FY95 accessions required to contract in FY94 % SHORT = difference between % REQUIRED and % ACHVD

LIST OF REFERENCES

- 1. USAREC MANUAL 100-5, *RECRUITING*, United States Army Recruiting Command, Fort Sheridan, Illinois, 1 Sep 89.
- 2. Information provided to the author by Hershberger, John, (Chief of Mission Division, PAE, USAREC), personal conversation.
- 3. Walpole, R.H. Myers. Probability and Statistics for Engineers and Scientists. New York: Macmillan Publishing Company, 1985.
- 4. Barlow, D.J. Bartholomew, J.M. Bremner, H.D. Brunk. Statistical Inference under Order Restrictions. John Wiley & Sons LTD, 1972.
- 5. Anthony Brooke, David Kendrick, Alexander Meeraus. GAMS, A USER'S GUIDE. The Scientific Press, 1992.

INITIAL DISTRIBUTION LIST

1.	Defense Technical Information Center Cameron Station Alexandria, VA 22304-6145	No .	Copies 2
2.	Library, Code 052 Naval Post Graduate School Monterey, CA 93943-5002		2
3.	Dr. Siriphong Lawphongpanich (OR/La) Department of Operations Resea: h Monterey, CA 93942-5004		2
4.	Dr. Lyn Whitaker (OR/Wh) Department of Operations Research Monterey, CA 93942-5004		1
5.	CDR, U.S. Army Recruiting Command ATTN: USARC-PAE-MSN (MAJ Hershberger) Ft. Knox, KY 40121		2
6.	CDR, U.S. Army Recruiting Command ATTN: USARC-RA (COL Engelage) Ft. Knox, KY 40121		1
7.	Dr. Edward Schmitz CNRC, Code 222C 4015 Wilson Blvd Arlington, VA 22203-1991		1
8.	CPT Bryan D. Burris 250 Willingham Dr Bonaire, GA 31005		3

j